PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (http://bmjopen.bmj.com/site/about/resources/checklist.pdf) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	Developing an evidence assessment framework and appraising the academic literature on migrant health in Malaysia: a scoping review
AUTHORS	de Smalen, Allard; Chan, Zhie; Abreu Lopes, Claudia; Vanore, Michaella; Loganathan, Tharani; Pocock, Nicola S.

VERSION 1 – REVIEW

REVIEWER	Anne MacFarlane	
	University of Limerick, Ireland.	
REVIEW RETURNED	28-Jul-2020	

GENE RAL COMM ENTS

This is an excellent scoping review, which is really well-written. It was most interesting to learn about the field of migrant health research in Malaysia. I believe that it will certainly enable researchers to address gaps in knowledge to improve the evidence base on migrant health in Malaysia in order to support policymakers with high quality evidence for decision-making

The paper has several strengths:

The authors pay careful attention to the issue of defining migrants and they explain their meaning of, and focus on, 'vulnerability'. This is not always the case in the field of migrant health so it is important to acknowledge the authors in this regard.

Apart from the fact that inter-rater reliability was limited to a 20% sample of the records in the first screening stage, the review is methodologically robust, following PRISMA and JBI guidelines for example. It makes methodological contributions by creating an evidence assessment framework, including modified Joanna Briggs Institute (JBI) checklists and a decision tree that identifies the type of study design and corresponding level of evidence.

The use of the Bay Area Regional Health Inequities Initiative (BARHII) framework as a conceptual framework was valuable and the framework was clearly explained and used.

The description and synthesis of findings in the Results and Discussion sections is meticulous, clear and engaging.

In terms of revisions, I have some comments for the authors' attention:

The Sweileh et al. 2018 paper is an important reference included in the Introduction. It has a focus on global migrant health research and highlights a lack of empirical studies about labour migrants. The findings of this review, therefore, are important given that the majority of studies were about this population. It would be good to see an explicit discussion about these findings from Malaysia and the Swelieh et al, 2018 paper in the Discussion

Other countries have seen the value of scoping review e.g. Ireland (reference below) and it would be valuable to include this in the Introduction so readers can gain a sense of similar work in the field

https://bmcpublichealth.biomedcentral.com/articles/10.1186/s12889-019-6651-2

The authors explain that the rationale for the selection of the BARHII framework is that it was

comprehensive and inclusive of various health dimensions, whereas most other models focused on specific public health elements or lacked clear explanation regarding the included

health-related components of the model. Please include some references for these other models.

The meaning of the following sentence is not clear to me because it sounds like the authors are seeing the BARHII framework's emphasis on health inequities as a limitation of some kind?

"Although the framework was developed to address health inequities, its coherent structure lends itself to this project as it allows the researchers to approach health from different perspectives".

The first paragraph in the Discussion makes excellent points about the need for migrant health governance, more epidemiological studies and attention to inter-sectoral contexts and collaboration. WHO Europe produced the first ever WHO Strategy and Action Plan for Refugee and Migrant Health in 2016 and makes important points like this. The authors may wish to include a reference:

https://www.euro.who.int/en/about-us/governance/regional-committee-for-europe/past-sessions/66th-session/documentation/working-documents/eurrc668-strategy-and-action-plan-for-refugee-and-migrant-health-in-the-who-european-region

A final comment, which does not require a response or action by the authors: I would like to point out that, while I understand that there were no patients involved in this study, there is still scope for Public and Patient Involvement in literature reviews. Further, it is important for researchers working in the field of migrant health to consider ways of increasing the involvement of migrants in our research. Here are some relevant references, which may be of interest to the authors for future projects: https://www.researchgate.net/publication/327838775_Patient_and_Public_Involvement_i n_Systematic_Reviews_A_Systematic_Review_of_Literature

https://bjgp.org/content/69/682/255/tab-article-info

REVIEWER	Nirmal Aryal
	Bournemouth University, UK
REVIEW RETURNED	17-Aug-2020

REVIEW RETURNED	17-Aug-2020	
GENERAL COMMENTS	This is an interesting piece of work stemmed from rigorous process. Malaysia has been one of the major destination countries for the migrants particularly from South Asia and South East Asia and I strongly believe that this review will improve the understanding, extent and scope of migration health research in Malaysia. Thank you for this extensive work. Following are my suggestions:	
	1. Introduction	
	 Line 64: Use the current global estimate of migrant workers which is 272 million according to the World Migration Report 2020. Try to minimize the use of definition of the terms like refugees, asylum seekers, regular or irregular migrants etc. You can provide the links to the reference if you wish. Line 105: All foreign workers are not vulnerable to health risk, please consider specifying the sub-groups like low-skilled labour migrants. 	

2. Methods:

- Methods section is comprehensive and well described.
- Line 142: Give reference to Bay Area Regional Health Inequities Initiative (BARHII) framework if possible.
- Line 149: patient and public involvement usually means how the potential beneficiaries (i.e. researchers for this review) were discussed and consulted for this review.
- Line 262: Description of level of evidence is repeated and also I think it best fits in line 250-51.

Results:

- Line 425 to 428, repeated from methods.
- Results on level of evidence could be summarized because associated table is self-explanatory.
- It would be interesting to understand main source of research on migration health in Malaysia (e.g.government institutions, public/private national academic institutions of Malaysia, international academic institutions, international organisations such as WHO or IOM etc). It would also useful to know if there are any discernible pattern of migration health research targeted to specific origin country (or countries) of the migrants or migrants from which countries are more/less studied.

Discussion

- Line 602, expatriates were not included in the review, it is not clear how expatriates were defined and how they differ from foreign workers.

Thank you.

REVIEWER	Daniel Vujcich	
	Curtin University, Australia	
REVIEW RETURNED	21-Aug-2020	

GENERAL COMMENTS

Thank you for the opportunity to review this paper. Your study design is rigorous and your attention to detail is to be commended. My specific comments (below) are offered to help improve the impact of your paper and ought not to detract from your sense of achievement in having undertaken this interesting and important study.

Introduction

- The paragraph commencing line 87 is very 'definition heavy'. It may be more appropriate to paraphrase the definitions here and set them out more fully in the methods section (e.g. at line 167).
- The reference used to support the statement "the total number of migrant workers is estimated to fall" was published in 2015. Is there a more contemporary source available to support this proposition?
- The introduction would benefit from a greater explanation as to why it is necessary to collate and summarise all publications since 1965. What is the relevance of such a broad retrospective review, as opposed to a review that is more focussed on recent studies (e.g. since 2000)?

Method

- Line 140: Consistent with PRISMA-ScR guidelines, please specify how the protocol can be accessed (e.g. on request from the first-named author).
- Line 146: "data was" should be "data were"

- Line 191: it may be useful to explain the logic behind these choices of databases, especially EconLit
- Line 193: It may be necessary to explain why the databases were searched from date of inception notwithstanding that studies were excluded if they were conducted or included data from 1965 or earlier.
- Line 202: it is stated that studies were included if they were conducted in Malaysia (it is not clear to me whether this includes cross-national studies in which Malaysia was included I assume so but please specify if not)
- Line 207: the decision to include only English studies (and not papers published in Bahasa Malaysia) seems very significant and the logic for imposing this criterion should be stated very clearly.
- Line 209: the sixth inclusion criteria is "inclusion of (im)migrants, foreign workers" etc (it is not clear to me whether this means that the review included general population surveys that may have included but not been directly aimed at migrants)
- Line 230: AS screened all articles and DC screened a 20% sample. According to Figure 3, discrepancies were found in approximately 13% of that sample. This should be explicitly stated in the text together with a brief description of the nature of the discrepancies.
- Line 232: Data were extracted by one reviewer. Were any steps taken to cross-check the accuracy of data extraction? If not, this should be stated as a limitation.
- Figure 2: it is not clear how the decision tree is applied in the case of mixed methods studies

Results

• Consideration could be given to other ways of presenting the data. The tables are very useful but extremely detailed and could be complemented by appropriate use of graphs (e.g. showing publications over time, proportion of publications by BARHII dimension etc).

Discussion

• The discussion makes statements about 'research gaps' based on studies spanning a very long period of time - 1965 to 2019. However, I think the discussion could be more nuanced. For instance, the fact that a particular area of research accounted for a low proportion of overall included studies may not necessarily be problematic. To demonstrate my point, as a contemporary sexual health researcher, the fact that (say) STI studies accounted for only 10% of the literature published since 1965 may not concern me if I knew that STI studies accounted for 50% of studies published in the last 10 years. Research gaps need to be interpreted in the context of shifting demographics, disease burdens etc.

REVIEWER	Gayle Halas University of Manitoba, Canada
REVIEW RETURNED	30-Aug-2020

GENERAL COMMENTS	BMJ Open: Developing an evidence framework and assessing the evidence on migrant health research in Malaysia: a scoping review	
	Manuscript copy has a number of areas flagged for further authors' review (see highlighted)	

<u>Abstract</u>

- databases searched would be a good addition
- I question whether this makes a methodological contribution yet? Not sure this is best stated in the first line of the conclusion.

Article Summary

- P 2 Authors' line 29-30: <u>fitting</u> the articles in to BARHII framework? If so, what criteria were used to exclude? Consider rewording as BARHII providing the framework used to extract and examine the articles.
- p 3 authors' line 52: consider changing "evidence base" to "research"
- Pg 3 Authors' Line 62 change "could" to "may"
- P 3 Authors' line 48: Would not consider this methodological research (Key word); scoping review would be a useful KW

Article purpose

- It is not clear why such a focus on quality and evidence is needed if simply scoping the research on [the nature and extent of] migrant health in Malaysia.
- I felt the word STATUS and ESTABLISH were not the most useful words to use in your objective, ie., "to explore the current status? and composition of migrant health research in Malaysia, and to establish? a framework to assess the quality of this academic literature." P 24 line 502 this statement is inconsistent with your stated study purpose but in fact is probably a more relevant way of stating the purpose of your review.

Outcomes

- Findings from the studies are well described. Few wording changes suggested. Would also recommend a different presentation of table 3....would be helpful to rank order the articles according to highest quality perhaps? Or according to the BARHII dimensions?
- Further consideration needed re quality viz. level of evidence.

Statistics

- A multiple-correspondence analysis (MCA)?? This is a new albeit very interesting approach for me. It would have been beneficial to describe how the points are plotted on the graph, and more appropriately label the axes.
- Again, not my area of expertise: how is the MCA affected by the size of the sample in each study and the number of studies?
- MCA: Why are different study designs used rather than the "levels of evidence" being proposed in this work?

Other comments

- Throughout the first section of the paper, I was uncertain of the scope of this work. It was very helpful to have the definitions provided, but might be strengthened even more by explicitly, up front, stating the term you will use to define your population of interest, and then stick to using that term thereafter. In my mind the pop of interest is migrant individuals in Malaysia who may be experiencing health-related vulnerabilities.
- Methods: P 8 Authors' line 167: what exactly are unclassified migrants is this the same as undocumented? (ie., different terminology being used comes across as confusing)
- Pg 9: For search strategy (particularly as provided in supplementary material) -- Why a list of conditions? Leaves the question of how these were determined or if it is complete (for example, "heart" is not specified). However, wider encompassing terms such as ailment*, health OR healthcare, morbidit* OR mortalit*, well-being OR wellbeing are present.

A decision tree was developed to classify the type of study design and level of evidence of each journal article. Did the author with expertise in EBM have significant input into the development of the decision tree?

Related concerns: Figure 2: <u>subjective</u> component regrading the decision of level 2 or 3 for qualitative studies. As stated in the footnote for Table 2: "Rigor was subjectively assessed and based on the number of included participants, amount of collected data, and detailed explanation how the study was conducted. " Not completely consistent with how trustworthiness is determined or conceptualized by qualitative methodologists.

Reliability: how did AS determine which BARHII health dimension was the focus or 'most prominent' within an article (as per PRISMA guideline for data charting process: describe...processes for obtaining and confirming data from investigators.) p 14 line 307

- P11 Authors line 232: data extracted and quality assessed by one reviewer weakens reliability
- P 12 Authors line 268 more detail needed to explain "incorporating feedback and multiple testing rounds"
- P 13 Lines 286-287 how are these scoring bands determined?
- P 13 Lines 293-5 how was the quality score calculated for <u>variables</u> versus articles?
- P 14 line 310 perhaps change "existing" to 'strength of

Results section: some very interesting findings presented here.

• P 18 line 383 – "foreign workers were with 36 articles..."-wording to be changed for easier reading. Line 388 – remove "merely" descriptive...makes it sound like weak research.

- P20 Line 439 its not the QUALITY of the evidence base....you are just reporting on a snapshot of research based on a determination of evidence LEVELS.
- P23 Line 447 please clarify what you mean by "non-appraised" article. (your table provides a better description)
- P 21 Lines 466+ re study quality by study design.... this decontextualized information (ie not attached to the articles themselves) felt less useful.
- P22 Line 475 I don't think you mean research dimension; consider research design or approach
- P 24 Line 511 wording issue: what are "health outcomes of other dimensions"?

Line 517-519: Non -health policies....countries" needs to be clarified, particularly a systematic review [of what?] in high income countries?

- P 26 Line 554 not just smoking behaviour. I think a more encompassing statement could be made about further research re risk behaviours and outcomes more generally.
- P 26 Line 563, as a first line in a paragraph, I'm not clear on which "most studies" you are referring to?
- Line 565-566; Not clear what is meant by "the latter issue could have occurred due to lacking information on the data used"
- P 27 Line 571-573: I suppose it is not surprising there are no studies on pops affected by illegal behaviour....perhaps change this to something like "...yet there is very little known about the health issues experienced by trafficked persons...."
- P 27 line 586-7: the issue isn't the quality of the evidence base or the need to produce papers, but that there is a need to conduct research that will provide strong evidence to support practices and policies that will positively impact migrant health.
- Pg 28 Line 595: not <u>all</u> identified peer-reviewed articles, but those that met the review inclusion criteria.
- Lines 598-600 wonder if this might be reworded to "we utilized a modified JBI toolkit and decision tree to help identify..., and found it to be useful for...." Similarly for p 31 lines 652-655: it would helpful to keep this open and invite people to use/modify this approach for enhancing understanding of the evidence base...etc....
- Tables 2 and 5: levels of evidence (from 1-4) are methodologically different in each design category so not sure the numbering system is suitable. Again requires review/input from an EBM expert.

•	Would be helpful to know how the JBI tools were modified.
•	PRISMA checklist appended but not completed.

VERSION 1 – AUTHOR RESPONSE

Response to Reviewer 1 Comments

Point 4: The Sweileh et al. 2018 paper is an important reference included in the Introduction. It has a focus on global migrant health research and highlights a lack of empirical studies about labour migrants. The findings of this review, therefore, are important given that the majority of studies were about this population. It would be good to see an explicit discussion about these findings from Malaysia and the Swelieh et al, 2018 paper in the Discussion

Response 4: Thank you for the suggestion. We now discuss findings in relation to Swelieh's 2018 papers (on global migration health research outputs, and migrant worker health outputs respectively), in the Discussion.

Page 27, line 541-544

Despite the strong representation, over half the research papers concentrated on communicable diseases, while only a few examined non-communicable diseases, consistent with global research output on international migrant workers.¹¹⁴

Page 27, line 548-552

We found that the majority of studies involved foreign workers (n=41/67), and only 10 studies examined asylum seekers and refugees as the primary population of interest. Our findings, therefore, offer useful synthesis on migrant worker's health specifically, which is lacking relative to studies on asylum seekers and refugees in global migration health research.¹⁴

Point 5: Other countries have seen the value of scoping review e.g. Ireland (reference below) and it would be valuable to include this in the Introduction so readers can gain a sense of similar work in the field

https://bmcpublichealth.biomedcentral.com/articles/10.1186/s12889-019-6651-2

Response 5: We have incorporated the feedback.

Page 6, line 103-105

Scoping reviews can be helpful to map the academic literature and have been used by different researchers to present the available evidence on migrant health in other countries.¹⁵

Point 6: The authors explain that the rationale for the selection of the BARHII framework is that it was comprehensive and inclusive of various health dimensions, whereas most other models focused on specific public health elements or lacked clear explanation regarding the included health-related components of the model. Please include some references for these other models.

Response 6: Other models are, for example, the Rural Community Health and Well-being Framework and Public Health Outcomes Framework. We have added the references for these models to this section.

Page 8-9, line 136-139

The BARHII framework was selected due to its comprehensive nature and inclusion of various health dimensions, whereas other models focused on specific public health elements or lacked clear explanation regarding the included health-related components of the model.²⁰

Point 7: The meaning of the following sentence is not clear to me because it sounds like the authors are seeing the BARHII framework's emphasis on health inequities as a limitation of some kind? "Although the framework was developed to address health inequities, its coherent structure lends itself to this project as it allows the researchers to approach health from different perspectives".

Response 7: We have removed this sentence, as this information does not improve the article's clarity.

Point 8: The first paragraph in the Discussion makes excellent points about the need for migrant health governance, more epidemiological studies and attention to inter-sectoral contexts and collaboration. WHO Europe produced the first ever WHO Strategy and Action Plan for Refugee and Migrant Health in 2016 and makes important points like this. The authors may wish to include a reference:

https://www.euro.who.int/en/about-us/governance/regional-committee-for-europe/past-sessions/66th-

<u>session/documentation/working-documents/eurrc668-strategy-and-action-plan-for-refugee-and-migrant-health-in-the-who-european-region</u>

Response 8: We have added the reference to the sentence on page 25, lines 494 to 497.

Among the five BARHII health dimensions, institutional inequities, and mortality and morbidity were the least represented. Yet, studies concerning the influence of governance on migrant health are of utmost importance, as overarching governance can affect health outcomes of the other BARHII dimensions. 99 100

Response to Reviewer 2 Comments

Point 9: Line 64. Use the current global estimate of migrant workers which is 272 million according to the World Migration Report 2020.

Response 9: We have made the recommended changes and added the World Migration Report 2020 reference.

Page 5, line 63-64

Worldwide, the international migrant population accounts for approximately 272 million people, with almost one-third within Asia.¹

Point 10: Try to minimize the use of definition of the terms like refugees, asylum seekers, regular or irregular migrants etc. You can provide the links to the reference if you wish.

Response 10: We have removed most of the in-text definitions and created a definitions table (please, see Table 1 in the manuscript).

Page 5, line 68-76

DOSM defines a non-citizen as a person that resides in Malaysia for six months or more in the reference year.⁴ However, no subcategories were included in this definition. According to the Office of the United Nations High Commissioner for Human Rights (OHCHR), a non-citizen is an individual that does not have an effective connection with the location where the person is situated according to the host nation, and includes various types of migrants, such as foreigners with permanent residency, refugees, asylum seekers, foreign labour, international students, stateless individuals, and victims of human trafficking.⁵ Other definitions of migrant-related terms that are used in this paper are presented in Table 1.

Point 11: Line 105. All foreign workers are not vulnerable to health risk, please consider specifying the sub-groups like low-skilled labour migrants.

Response 11: We have made the recommended changes.

Page 6, line 94-96

Refugees, asylum seekers, and both documented and undocumented low-skilled foreign workers can be classified as vulnerable migrants in Malaysia, as these populations may face significant hardships in their new country of residence.⁹ 10

Point 12: Line 142. Give reference to Bay Area Regional Health Inequities Initiative (BARHII) framework if possible.

Response 12: We have added the reference.

Page 8, line 122-123

Data were extracted and organised using the Bay Area Regional Health Inequities Initiative (BARHII) framework.¹⁸

Point 13: Line 149. Patient and public involvement usually means how the potential beneficiaries (i.e. researchers for this review) were discussed and consulted for this review.

Response 13: Thank you for the clarification. We have made the following changes:

Page 8, line 130-132

There were no patients involved in this study. The findings of this study were presented at the Migrant Health Research Dissemination Workshop in Kuala Lumpur to stakeholders working on migrant health in Malaysia.¹⁹

Point 14: Line 262. Description of level of evidence is repeated and also I think it best fits in line 250-51.

Response 14: Although the information concerning the level of evidence was not repeated, we have made some changes to clarify this section.

Page 13, line 240-251

The decision tree built on the study design tree from the Centre for Evidence-Based Medicine (CEBM)²⁴ and essentially allowed research of varying designs to be consistently, reliably classified into one of several design families. The newly developed decision tree was created through a two-step process. First, a definitions table of included research designs was developed to adapt specific characteristics of each definition into the decision tree to identify the paper's study design (Table 2).

[INSERT TABLE 2]

Second, Tomlin & Borgetto's²⁹ model was utilised to identify the level of evidence of the included literature, as the study designs that were included in their model were in line with the research designs in the definitions table.

Point 15: Line 425 to 428, repeated from methods.

Response 15: We have removed this part.

Point 16: Results on level of evidence could be summarized because associated table is self-explanatory.

Response 16: We have made the recommended changes and summarised this section.

Page 22-23, line 455-466

In general, the quality of the evidence base on migrant health in Malaysia is low (49.2%) and consists mostly of level 3 evidence papers (n=27/65). Level 2 evidence represents 38.5% of the evidence base (n=25/65), followed by level 4 evidence papers (n=13/65). No level 1 evidence studies (systematic reviews or meta-analyses) were identified. The majority of the papers (n=41/65) focused on foreign workers, however, studies that included asylum seekers and refugees have the highest mean quality (58.4%). Furthermore, only four out of five BARHII health dimensions were included in the quality assessment. The living conditions dimension has the highest average score (59.7%), followed by the risk behaviour dimension (48.7%), mortality and morbidity dimension (47.9%), and the disease and injury dimension (46.3%). Moreover, the descriptive research category represents the majority (70.8%) of the evidence base with a mean quality of 47.7%. The qualitative research category has the highest mean quality and is the only research category with a high-quality score (76%).

Point 17: It would be interesting to understand main source of research on migration health in Malaysia (e.g. government institutions, public/private national academic institutions of Malaysia, international academic institutions, international organisations such as WHO or IOM etc). It would also useful to know if there are any discernible pattern of migration health research targeted to specific origin country (or countries) of the migrants or migrants from which countries are more/less studied.

Response 17: Thank you for the suggestion. As with other countries, migration health research is comprised of a mix of the sources above - there were no discernible patterns by source, but as noted in the Limitations section, only peer-reviewed literature was searched, which may have missed some relevant grey literature from UN agencies and NGOs. We also did not find any specific patterns of investigation by migrant origin country. Given these findings, we do not think it relevant to include this as a discussion point in the paper.

Point 18: Line 602, expatriates were not included in the review, it is not clear how expatriates were defined and how they differ from foreign workers.

Response 18: We have added some examples to show the difference between expatriates and foreign workers.

Page 29, line 589-594

As this paper focuses exclusively on vulnerable migrants within the non-citizen population in Malaysia, we excluded other non-citizen groups, such as expatriates and international students, based on the assumption that these groups are less vulnerable (e.g., expatriates in Malaysia have more privileges in terms of recognition regarding their roles in society, receive better financial compensation, and tend to have access to many other benefits compared to foreign workers).

Response to Reviewer 3 Comments

Point 19: The paragraph commencing line 87 is very 'definition heavy'. It may be more appropriate to paraphrase the definitions here and set them out more fully in the methods section (e.g. at line 167).

Response 19: Please, see 'Response 10.'

Point 20: The reference used to support the statement "the total number of migrant workers is estimated to fall" was published in 2015. Is there a more contemporary source available to support this proposition?

Response 20: We have changed the old reference to a reference from the IOM (2020)

Page 5, line 83-84

However, the total number of migrant workers, both documented and undocumented, is estimated to fall between 4.2 and 6.2 million people.²

Point 21: The introduction would benefit from a greater explanation as to why it is necessary to collate and summarise all publications since 1965. What is the relevance of such a broad

retrospective review, as opposed to a review that is more focussed on recent studies (e.g. since 2000)?

Response 21: Thank you for the suggestion. We have decided not to give an explanation for the year cut-off (1965) in the introduction, but in the methods section.

Page 11, line 204-206

Studies were excluded if they were: 1) conducted or included data from 1965 or earlier, as Singapore was part of Malaysia until 1965, and this study is careful to only include Malaysia studies without Singapore; ...

Point 22: Line 140. Consistent with PRISMA-ScR guidelines, please specify how the protocol can be accessed (e.g. on request from the first-named author).

Response 22: We have incorporated the feedback.

Page 8, line 121-122

The pre-review protocol can be accessed on request from the first author.

Point 23: Line 146, "data was" should be "data were"

Response 23: We have incorporated the feedback.

Page 8, line 122-123

Data were extracted and organised using the Bay Area Regional Health Inequities Initiative (BARHII) framework.¹⁸

Point 24: Line 191. It may be useful to explain the logic behind these choices of databases, especially EconLit

Response 24: We have incorporated the feedback and added some references to this section.

Page 10, line 176-179

Based on the guidelines of the London School of Hygiene and Tropical Medicine²² and Bramer et al²³ on selecting the number and types of databases that should be included in biomedical systematic searches, six databases were selected for this study: Econlit, Embase, Global Health, Medline, PsycInfo, and Social Policy and Practice.

Point 25: Line 193. It may be necessary to explain why the databases were searched from date of inception notwithstanding that studies were excluded if they were conducted or included data from 1965 or earlier.

Response 25: We have incorporated the feedback.

Page 10, line 179-182

This scoping review includes studies from 1965 onwards until 2019. However, all records (including records published before 1965) were retrieved to manually screen the data for publication date-related issues.

Point 26: Line 202. It is stated that studies were included if they were conducted in Malaysia (it is not clear to me whether this includes cross-national studies in which Malaysia was included – I assume so but please specify if not)

Response 26: We have incorporated the feedback.

Page 11, line 192-193

Studies were eligible for inclusion if they met the following inclusion criteria: 1) conducted in Malaysia, including cross-national studies in which Malaysia was included; ...

Point 27: Line 207. The decision to include only English studies (and not papers published in Bahasa Malaysia) seems very significant and the logic for imposing this criterion should be stated very clearly.

Response 27: This decision is based on the fact that the first author does not speak the Malaysian language. Therefore, he would not have been able to verify the data from the second reviewer. However, only one paper in Bahasa Malaysia was identified and is mentioned in the limitations section.

Page 29, line 600-601

Also, only English language articles were included, resulting in the exclusion of one paper in Bahasa Malaysia (the Malay language). 120

Point 28: Line 209. The sixth inclusion criteria is "inclusion of (im)migrants, foreign workers" etc (it is not clear to me whether this means that the review included general population surveys that may have included – but not been directly aimed at - migrants)

Response 28: We have incorporated the feedback.

Page 11, line 199-203

... 6) inclusion of international (im)migrants, foreign workers, asylum seekers, and refugees, as these groups were considered as vulnerable migrant populations in Malaysia. Articles that included both migrants and the general population were included in this study if sufficient information concerning the migrant population was available.

Point 29: Line 230. AS screened all articles and DC screened a 20% sample. According to Figure 3, discrepancies were found in approximately 13% of that sample. This should be explicitly stated in the text together with a brief description of the nature of the discrepancies.

Response 29: We have incorporated the feedback.

Page 12, line 224-226

In both stages, the discrepancies were about 13% to 14% of the papers and were mostly around the study design and target populations. Conflicts were examined and resolved by NP.

Point 30: Line 232. Data were extracted by one reviewer. Were any steps taken to cross-check the accuracy of data extraction? If not, this should be stated as a limitation.

Response 30: We mentioned this issue in the limitations section. However, we have incorporated an additional statement concerning this matter in the methods section to improve the transparency of this article.

Page 13, line 231-232

Data extraction and categorisation into the BARHII framework categories was not cross-checked by a second reviewer due to time and human resource constraints.

Page 30, line 604-606

Inter-rater reliability was limited to a 20% sample of the records in the first screening stage, and no data extraction nor quality assessment was verified by a second reviewer due to resource constraints.

Point 31: Figure 2: it is not clear how the decision tree is applied in the case of mixed methods studies

Response 31: We have incorporated the feedback.

Page 13, line 235-236

Although various research designs were included in the decision tree, some study designs did not fit in this model, such as the mixed-method design.

Point 32: Consideration could be given to other ways of presenting the data. The tables are very useful but extremely detailed and could be complemented by appropriate use of graphs (e.g. showing publications over time, proportion of publications by BARHII dimension etc).

Response 32: We have created three additional figures (bar charts) to present overviews of the number of studies disaggregated by health dimension, type of migrant, and research design (please, see Figures 4-6 in the manuscript).

Page 18, line 353-354

Figures 4 and 5 present overviews of the number of studies disaggregated by health dimension and type of migrant, respectively.

Page 22, line 446-447

In addition, Figure 6 shows an overview of the number of studies disaggregated by research design.

Point 33: The discussion makes statements about 'research gaps' based on studies spanning a very long period of time - 1965 to 2019. However, I think the discussion could be more nuanced. For instance, the fact that a particular area of research accounted for a low proportion of overall included studies may not necessarily be problematic. To demonstrate my point, as a contemporary sexual health researcher, the fact that (say) STI studies accounted for only 10% of the literature published since 1965 may not concern me if I knew that STI studies accounted for 50% of studies published in the last 10 years. Research gaps need to be interpreted in the context of shifting demographics, disease burdens etc.

Response 33: Thank you for this interesting point. This review catalogues the breadth and quality of migrant health research in Malaysia since the beginnings of nationhood to the present day, which may not take into account shifting demographics, disease burdens and corresponding research priorities. Therefore, future research could focus on subsequent analyses of this dataset to explore these nuances.

Response to Reviewer 4 Comments

Point 34: Abstract – Databases searched would be a good addition.

Response 34: Please, see 'Response 1.'

Point 35: I question whether this makes a *methodological* contribution yet? Not sure this is best stated in the first line of the conclusion.

Response 35: We have removed this part from the conclusion in the abstract.

Point 36: Page 2 Authors' line 29-30. Fitting the articles in to BARHII framework? If so, what criteria were used to exclude? Consider rewording as BARHII providing the framework used to extract and examine the articles.

Response 36: We have reworded this section.

Page 2, line 28-31

Studies were eligible for inclusion if they were conducted in Malaysia, peer-reviewed, focused on a health dimension according to the Bay Area Regional Health Inequities Initiative (BARHII) framework, and targeted the vulnerable international migrant population.

Point 37: Page 3 authors' line 52. Consider changing "evidence base" to "research."

Response 37: We have changed "evidence base" to "research."

Page 4, line 52-54

This study provides a comprehensive overview of migrant health research in Malaysia, including a summary table, critical assessment tables, and a multiple-correspondence analysis (MCA).

Point 38: Page 3 Authors' Line 62. Change "could" to "may."

Response 38: We have changed "could" to "may."

Page 4, line 59-61

Only English peer-reviewed academic articles were included in this study, and, therefore, much relevant information that could potentially be used to inform both policies and practice may have been excluded from this review.

Point 39: Page 3 Authors' line 48. Would not consider this methodological research (Key word); scoping review would be a useful KW Article purpose

Response 39: We have changed the key word "methodological research" to "evidence assessment framework."

Page 3, lines 48-49

Key Words: Malaysia, migrant, health, refugee, foreign worker, disease, evidence assessment framework

Point 40: It is not clear why such a focus on quality and evidence is needed if simply scoping the research on [the nature and extent of] migrant health in Malaysia.

Response 40: In scoping reviews, quality appraisal according to PRISMA-ScR guidelines, ¹⁷ is optional, as long as authors state the rationale for doing so. We provide this rationale in the Introduction:

Page 6-7, line 109-113

Aggravating the matter, there is no overall picture currently available of the evidence base on migrant health in Malaysia, including critical appraisal of the quality of research. Therefore, this study aims to map the existing academic literature on migrant health in Malaysia since 1965 to identify the gaps in this field, as well as to present an overview of the quality and level of evidence of these scientific studies.

Point 41: I felt the word STATUS and ESTABLISH were not the most useful words to use in your objective, ie., "to explore the current status? and composition of migrant health research in Malaysia, and to establish? a framework to assess the quality of this academic literature." P 24 line 502 – this statement is inconsistent with your stated study purpose – but in fact is probably a more relevant way of stating the purpose of your review.

Response 41: We have incorporated the feedback and reworded this section.

Page 2, line 24-26

This study aims to map the existing academic literature on migrant health in Malaysia and to provide an overview of the quality and level of evidence of these scientific studies.

Point 42: Findings from the studies are well described. Few wording changes suggested. Would also recommend a different presentation of table 3....would be helpful to rank order the articles according to highest quality perhaps? Or according to the BARHII dimensions?

Response 42: Thank you for the suggestion. However, we will keep this order, as the table is organised from oldest article first to latest article last.

Point 43: Further consideration needed re quality viz. level of evidence.

Response 43: We present findings on mean study quality by study design, in relation to the level of evidence in Table 6., with discussion of these findings on pages 22-23, line 442-466. Given the already long word count and extensive description of findings by BARHII dimension, we prefer not to expand this section further. In our view doing so would not add to the findings.

Point 44: Statistics – A multiple-correspondence analysis (MCA)?? This is a new albeit very interesting approach for me. It would have been beneficial to describe how the points are plotted on the graph, and more appropriately label the axes.

Response 44: The MCA is a latent analysis that summarises the relationship between the levels of categorical variables. It projects categories in a bi-dimensional space based on weighted Euclidean distances. The axes are statistical latent dimensions and, therefore, not possible to label.

We have also added more information to this section to describe the MCA better.

Page 16, line 309-322

An MCA is a descriptive technique that can be utilised to visually demonstrate relationships among the levels of several categorical variables – here, these include the type of migrant, main health dimension, quality of the study, and research design – in a two-dimensional space. The MCA projects categories in a two-dimensional space with axes defined by latent dimensions (and, therefore, it is not possible to label the axes), based on weighted Euclidean distances.³¹ The MCA allows categories with similar profiles to be grouped together, where a closer distance of categories within the same quadrant demonstrates a stronger relationship, whereas categories that are further apart and in opposite quadrants present weaker associations.³² In addition to the MCA, chi-square tests were conducted to assess whether categorical variables were independent (e.g., not associated). It should be noted that a few studies included two BARHII dimensions, yet, the analysis only allowed one dimension to be included. Therefore, only the most prominent dimension, based on the amount of attention given to the specific dimension in the article, was selected and used for the analysis.

Point 45: Statistics – Again, not my area of expertise: how is the MCA affected by the size of the sample in each study and the number of studies?

Response 45: The multiple correspondence analysis is a descriptive technique with no statistical assumptions. There is no significance test in the analysis, therefore the size of the sample will not affect the results of the analysis. However, larger sample sizes produce more robust results since the analysis is sensitive to variations in values.

Point 46: Statistics – MCA: Why are different study designs used rather than the "levels of evidence" being proposed in this work?

Response 46: The MCA requires categorical variables and 'levels of evidence' is an ordinal variable. However, Table 3 presents the correspondence between research designs and levels of evidence. The results of the analysis types of design can also be interpreted alluding to levels of evidence.

Point 47: Throughout the first section of the paper, I was uncertain of the scope of this work. It was very helpful to have the definitions provided, but might be strengthened even more by explicitly, up front, stating the term you will use to define your population of interest, and then stick to using that term thereafter. In my mind – the pop of interest is migrant individuals in Malaysia who may be experiencing health-related vulnerabilities.

Response 47: Please, see 'Response 10.'

Point 48: Page 8 Authors' line 167. What exactly are unclassified migrants – is this the same as undocumented? (ie., different terminology being used comes across as confusing)

Response 48: We have added an explanation to this section.

Page 9, lines 149 to 151

The lattermost category was applied if a paper used the term 'migrants' or 'immigrants' but lacked specific information to classify the study population as foreign workers or asylum seekers/refugees.

Point 49: Page 9. For search strategy (particularly as provided in supplementary material) -- Why a list of conditions? Leaves the question of how these were determined or if it is complete (for example, "heart" is not specified). However, wider encompassing terms such as ailment*, health OR healthcare, morbidit* OR mortalit*, well-being OR wellbeing are present.

Response 49: We have added an explanation to this section.

Page 10-11, line 182-190

The search process was conducted by AS and included a two-stage procedure to ensure that the search was exhaustive and to minimise the risk of missing potentially eligible studies. The first stage focused on identifying English-language key words and Medical Subject Headings (MeSH) terms for migrants (e.g., immigrants, foreign workers, refugees), health (e.g., disease, infection, disorder), and Malaysia (e.g., Sabah, Kuala Lumpur) through reading search strategies of other review studies on migrant health as well as looking over medical

terminology of renowned medical institutions, such as the Mayo Clinic. Subsequently, these items were combined by using Boolean operators (e.g., migrant AND health AND Malaysia) in the search platform of each database (Supplementary file 2).

Point 50: A decision tree was developed to classify the type of study design and level of evidence of each journal article. Did the author with expertise in EBM have significant input into the development of the decision tree?

Response 50: We used the CEBM framework as a starting point for creating the decision tree, followed by Tomlin & Borgetto's evidence-based pyramid model, as described in the Methods section (page 13-14, line 233-270). The lead author led decision tree development, supported by co-authors with expertise in systematic reviews and evidence frameworks.

Point 51: Related concerns: Figure 2: subjective component regarding the decision of level 2 or 3 for qualitative studies. As stated in the footnote for Table 2: "Rigor was subjectively assessed and based on the number of included participants, amount of collected data, and detailed explanation how the study was conducted. "Not completely consistent with how trustworthiness is determined or conceptualized by qualitative methodologists.

Response 51: Tomlin & Borgetto (2011) used the categories 'more rigor' and 'less rigor' in their model. Unfortunately, they did not describe how to objectively assess the level of rigor in a qualitative study, and only provided three subjective criteria: a) prolonged engagement with participants; b) triangulation of data (multiple sources); c) confirmation of data analysis and interpretation (peer and member checking). Therefore, we do admit that this is a limitation of the decision tree, and we would like to address this aspect in future research to improve our model.

Point 52: Reliability. How did AS determine which BARHII health dimension was the focus or 'most prominent' within an article (as per PRISMA guideline for data charting process: describe...processes for obtaining and confirming data from investigators.) p 14 line 307

Response 52: We have added an explanation to this section.

Page 16, line 318-322

It should be noted that a few studies included two BARHII dimensions, yet, the analysis only allowed one dimension to be included. Therefore, only the most prominent dimension, based on the amount of attention given to the specific dimension in the article, was selected and used for the analysis.

Point 53: Page 11 Authors line 232. Data extracted and quality assessed by one reviewer - weakens reliability.

Response 53: Please see 'Response 30.'

Point 54: Page 12 Authors line 268 – more detail needed to explain "incorporating feedback and multiple testing rounds."

Response 54: We have added some additional information to this sentence.

Page 14, line 267-270

After incorporating feedback on the questions used to identify the research design and multiple testing rounds to assess if the questions were specific enough to distinguish these designs within the full set of articles, the final version of the decision tree – as seen in Figure 2 – was used to extract the data.

Point 55: Page 13 Lines 286-287 – how are these scoring bands determined?

Response 55: We have added an explanation to this section.

Page 15, line 286-301

As the JBI toolkit has no standard scoring index, the following scoring system was applied: 1) low quality = 0% to 50%; 2) moderate quality = above 50% and below 75%; 3) high quality = 75% or higher. Although a four-band scoring system – where each category would include a 25% scoring range – was considered, a three-band scoring system was selected because the

three given categories – low, moderate, and high – would simplify the interpretation concerning the quality of the study. In a four-band system, the distinction and classification of the two middle categories are less straightforward compared to the three-band scoring system. Further, the first two categories in a four-band scoring system would still represent a poor-quality study, and, hence, should be used to signal more cautious interpretation of the study results among readers. The cut-off score was based on the idea that if a study could answer 'yes' to only half or less of the questions, it would not be sufficient to transmit a reliable message to the audience. Therefore, at least more than half of the questions should be answered with 'yes' to obtain a moderate score. The 75% cut-off was still based on the idea of having four equal scoring categories, where 75% and above would be classified as a high-quality study and would inform the audience with a more credible message.

Point 56: Page 13 Lines 293-5 – how was the quality score calculated for variables versus articles?

Response 56: We have added some information to clarify this point.

Page 16, line 305-307

Mean quality scores were calculated for the different variables by using Microsoft Excel, including the type of migrant, health dimension, health subdimension, research design, and level of evidence.

Point 57: Page 14 line 310 – perhaps change "existing" to 'strength of

Response 57: We have reworded this section, please see 'Response 44.'

Point 58: Page 18 line 383 – "foreign workers were with 36 articles..." - wording to be changed for easier reading.

Response 58: We have reworded this sentence.

Page 20, line 401-403

Most articles (n=36/46) studied foreign workers, ^{37 40 43 44 46-56 58-60 66 69 73-75 78 79 81 83-86 89 91-93 95 97} while only six and four articles included unclassified migrants ^{39 45 57 65 68 70} and refugee populations. ^{33 34 71 87} respectively.

Point 59: Line 388 - remove "merely" descriptive...makes it sound like weak research.

Response 59: We have removed the word "merely" in this sentence.

Page 20, line 406-407

Most of the studies were descriptive and presented that migrants, irrespective of the defined type, represented a significant share among the study populations.

Point 60: Page 20 Line 439 – its not the QUALITY of the evidence base....you are just reporting on a snapshot of research based on a determination of evidence LEVELS.

Response 60: We do talk about the quality of the evidence base, as we conducted critical appraisal to assess the quality of the articles. In addition, we assessed the level of evidence of the articles.

Point 61: Page 23 Line 447 – please clarify what you mean by "non-appraised" article. (your table provides a better description)

Response 61: We have removed the 'non-appraised' sentence and reworded this section.

Page 22, line 444-446

Two articles – representing a scoping review⁷² and mixed-method design⁸⁰ – were excluded from this assessment, as the JBI toolkit does not accommodate these study designs.

Point 62: Page 21 Lines 466+ re study quality by study design.... this decontextualized information (ie not attached to the articles themselves) felt less useful.

Response 62: Please, see 'Response 16'.

Point 63: Page 22 Line 475 – I don't think you mean research dimension; consider research design or approach.

Response 63: We have changed the word "dimension" to "category."

Page 23, line 464-466

Moreover, the descriptive research category represents the majority (70.8%) of the evidence base with a mean quality of 47.7%. The qualitative research category has the highest mean quality and is the only research category with a high-quality score (76%).

Point 64: Page 24 Line 511 – wording issue: what are "health outcomes of other dimensions"?

Response 64: We have reworded this sentence.

Page 25, line 493-496

Among the five BARHII health dimensions, institutional inequities, and mortality and morbidity were the least represented. Yet, studies concerning the influence of governance on migrant health are of utmost importance, as overarching governance can affect health outcomes of the other BARHII dimensions. ^{99 100}

Point 65: Line 517-519. Non-health policies....countries" needs to be clarified, particularly a systematic review [of what?] in high income countries?

Response 65: We have reworded this sentence.

Page 25, line 503-505

Furthermore, a recent systematic review on the effects of non-health-targeted policies on migrant health in high-income countries showed that non-health policies (e.g., restrictive immigration policies) were associated with poor health outcomes.¹⁰³

Point 66: Page 26 Line 554 – not just smoking behaviour. I think a more encompassing statement could be made about further research re risk behaviours and outcomes more generally.

Response 66: We have incorporated the feedback.

Page 27, line 537-539

Therefore, future research should further explore the differences in other risk behaviours, such as smoking, between Malaysians and migrants in Malaysia.

Point 67: Page 26 Line 563, as a first line in a paragraph, I'm not clear on which "most studies" you are referring to?

Response 67: We have reworded this sentence.

Page 27, line 548-549

We found that the majority of studies involved foreign workers (n=41/67), and only ten studies examined asylum seekers and refugees as the primary population of interest.

Point 68: Line 565-566. Not clear what is meant by "the latter issue could have occurred due to lacking information on the data used"

Response 68: We have reworded this sentence.

Page 27-28, line 552-556

Furthermore, eleven studies did not specify the included migrant population. This issue could have occurred due to missing information on the type of migrant in the dataset that the researchers used for their studies. For example, the Ministry of Health (MOH) will not report anything more detailed than 'non-Malaysian,' as no further information on non-citizens are collected during patient registration at MOH facilities.

Point 69: Page 27 Line 571-573: I suppose it is not surprising there are no studies on pops affected by illegal behaviour....perhaps change this to something like "...yet there is very little known about the health issues experienced by trafficked persons...."

Response 69: We have reworded this sentence.

Page 28, line 558-560

Also, human trafficking could significantly affect a person's health and vulnerability, yet, there is very little known about the health issues experienced by trafficked persons in Malaysia.¹¹⁶

Point 70: Page 27 line 586-7. The issue isn't the quality of the evidence base or the need to produce papers, but that there is a need to conduct research that will provide strong evidence to support practices and policies that will positively impact migrant health.

Response 70: We have reworded this sentence.

Page 28, line 573-575

Therefore, there is a clear need to conduct research that will provide strong evidence to support practices and policies that will positively impact migrant health.

Point 71: Page 28 Line 595. Not all identified peer-reviewed articles, but those that met the review inclusion criteria.

Response 71: We have incorporated the feedback.

Page 29, line 581-583

This study is the first systematic literature synthesis and scoping review on migrant health in Malaysia and presents a comprehensive overview of all identified peer-reviewed articles that met the inclusion criteria.

Point 72: Lines 598-600 – wonder if this might be reworded to "we utilized a modified JBI toolkit and decision tree to help identify..., and found it to be useful for...." Similarly for p 31 lines 652-655: it would helpful to keep this open and invite people to use/modify this approach for enhancing understanding of the evidence base...etc....

Response 72: We have incorporated the feedback.

Page 29, line 584-588

Furthermore, we utilised a self-developed decision tree and modified JBI checklists to help identify the type of study design and corresponding level of evidence of the included studies. We found this evidence assessment framework to be useful for the quality assessment of migrant health-related studies, and it might be useful for other research fields as well.

Point 73: Tables 2 and 5: levels of evidence (from 1-4) are methodologically different in each design category so not sure the numbering system is suitable. Again – requires review/input from an EBM expert.

Response 73: Table 2 (changed to Table 3, see page 54) and Table 5 (changed to Table 6, see page 72) are the same, however, Table 5 splits up 'case series' and 'prevalence study without analytical component' to make sure that every individual research design gets its own attention.

Point 74: Would be helpful to know how the JBI tools were modified.

Response 74: We have developed objective criteria for each question of the JBI checklists, as the previous criteria was rather subjective, and, therefore, could affect critical appraisal scores among researchers due to a different interpretation of the question. Also, we have slightly modified the sentence on this matter.

Page 14-15, line 275-279

Additional objective criteria specific to migrant health studies were developed for each question of the JBI checklists to increase the reliability of the quality assessment. After discussing the additional criteria and piloting the tools, slight modifications were made for the JBI tools, and these final versions were used to assess the quality of the papers. The modified checklists can be accessed on request from the first author.

Point 75: PRISMA checklist appended but not completed.

Response 75: The PRISMA checklist was completed, but it was attached as a separate file. Please see the PDF file (Supplementary file 1).

References

1. International Organization for Migration. World migration report 2020. Geneva: International Organization for Migration 2019.

- 2. International Organization for Migration. Malaysia [Internet]. Kuala Lumpur: International Organization for Migration; 2020 [cited 2020 Oct 01]. Available from: https://www.iom.int/countries/malaysia
- 3. Department of Statistics Malaysia. Population quick info [Internet]. Putrajaya: Department of Statistics Malaysia; 2020 [updated 2020 Apr 30; cited 2020 Apr 30]. Available from: https://www.dosm.gov.my/v1/index.php?r=column/cone&menu_id=b2RVUmEyZ2ZZYXhaUzUrdnpsa Wozdz09
- 4. Department of Statistics Malaysia. Current population estimates Malaysia 2019. Putrajaya: Department of Statistics Malaysia.
- 5. Office of the United Nations High Commissioner for Human Rights. The rights of non-citizens. Geneva: United Nations 2006.
- 6. International Organization for Migration. Glossary on migration. 2nd ed. Geneva: International Organization for Migration 2011.
- 7. Ministry of Home Affairs Malaysia. Statistik pekerja asing terkini mengikut negeri dan sektor [Internet]. Putrajaya: Ministry of Home Affairs Malaysia; 2019 [updated 2019 Aug 1; cited 2020 Apr 14]. Available from: http://www.data.gov.my/data/ms_MY/dataset/statistik-pekerja-asing-terkini-mengikut-negeri-dan-sektor
- 8. United Nations High Commissioner for Refugees. Figures at a glance in Malaysia [Internet]. Kuala Lumpur: United Nations High Commissioner for Refugees; 2019 [cited 2019 Nov 20]. Available from: https://www.unhcr.org/en-my/figures-at-a-glance-in-malaysia.html
- 9. Chuah FL, Tan ST, Yeo J, Legido-Quigley H. The health needs and access barriers among refugees and asylum-seekers in Malaysia: a qualitative study. International Journal for Equity in Health. 2018 Dec 1;17(1):120.
- 10. Devadason ES, Meng CW. Policies and laws regulating migrant workers in Malaysia: A critical appraisal. Journal of Contemporary Asia. 2014 Jan 2;44(1):19-35.
- 11. International Organization for Migration. IOM handbook: protection and assistance for migrants vulnerable to violence, exploitation and abuse. Geneva: International Organization for Migration 2018.
- 12. Pocock NS, Suphanchaimat R, Chan CK, Faller EM, Harrigan N, Pillai V, Wickramage K. Reflections on migrant and refugee health in Malaysia and the ASEAN region. BMC Proceedings. 2018 Apr;12(Suppl 4):4.
- 13. Lasimbang HB, Tong WT, Low WY. Migrant workers in Sabah, East Malaysia: The importance of legislation and policy to uphold equity on sexual and reproductive health and rights. Best Practice & Research Clinical Obstetrics & Gynaecology. 2016 Apr 1;32:113-23.
- 14. Sweileh WM, Wickramage K, Pottie K, Hui C, Roberts B, Sawalha AF, Zyoud SH. Bibliometric analysis of global migration health research in peer-reviewed literature (2000–2016). BMC Public Health. 2018 Dec;18(1):777.
- 15. Villarroel N, Hannigan A, Severoni S, Puthoopparambil S, MacFarlane A. Migrant health research in the Republic of Ireland: a scoping review. BMC Public Health. 2019 Dec;19(1):324.
- 16. Khanlou N, Haque N, Skinner A, Mantini A, Kurtz Landy C. Scoping review on maternal health among immigrant and refugee women in Canada: prenatal, intrapartum, and postnatal care. Journal of Pregnancy 2017 Jan;2017:1-14

- 17. Tricco AC, Lillie E, Zarin W, O'Brien KK, Colquhoun H, Levac D, Moher D, Peters MD, Horsley T, Weeks L, Hempel S. PRISMA extension for scoping reviews (PRISMA-ScR): Checklist and explanation. Annals of Internal Medicine. 2018 Oct 2;169(7):467-73.
- 18. Bay Area Regional Health Inequities Initiative. Framework [Internet]. Oakland, CA: Bay Area Regional Health Inequities Initiative; 2015 [cited 2019 Sep 18]. Available from: http://barhii.org/framework/
- 19. United Nations University International Institute for Global Health. Migrant Health Research Dissemination Workshop [Internet]. Kuala Lumpur: United Nations University International Institute for Global Health; 2020 Jan 31 [cited 2020 Oct 3]. Available from: https://iigh.unu.edu/news/news/policy-gaps.html
- 20. National Academies of Sciences, Engineering, and Medicine. A framework for educating health professionals to address the social determinants of health. Washington, DC: The National Academies Press 2016.
- 21. Public Health England. Public Health Outcomes Framework 2019/20: a consultation. London: Public Health England 2019.
- 22. London School of Hygiene & Tropical Medicine. Which databases should I search? A guide to the most popular resources. London: London School of Hygiene & Tropical Medicine 2014.
- 23. Bramer WM, Rethlefsen ML, Kleijnen J, Franco OH. Optimal database combinations for literature searches in systematic reviews: a prospective exploratory study. Systematic Reviews. 2017 Dec 1;6(1):245.
- 24. Centre for Evidence-Based Medicine. Study designs. Oxford: Centre for Evidence-Based Medicine; 2019 [cited 2019 Oct 10]. Available from: https://www.cebm.net/2014/04/study-designs/
- 25. Centre for Cognitive Ageing and Cognitive Epidemiology. Systematic reviews and metaanalyses: A step-by-step guide. Edinburgh: Centre for Cognitive Ageing and Cognitive Epidemiology; 2013 [cited 2019 Oct 15]. Available from: https://www.ccace.ed.ac.uk/research/softwareresources/systematic-reviews-and-meta-analyses
- 26. Alexander LK, Lopes B, Ricchetti-Masterson K, Yeatts KB. ERIC notebook [2nd ed]. Chapel Hill: Epidemiologic Research and Information Center, 2015.
- 27. Cochrane Community. Glossary. London: Cochrane Community; 2019 [cited 2019 Nov 2]. Available from: https://community.cochrane.org/glossary#letter-N
- 28. Peacock JL, Peacock PJ. Oxford Handbook of Medical Statistics. Oxford: Oxford University Press 2011.
- 29. Tomlin G, Borgetto B. Research pyramid: A new evidence-based practice model for occupational therapy. American Journal of Occupational Therapy. 2011 Mar 1;65(2):189-96.
- 30. Aromataris E, Munn Z, editors. Joanna Briggs Institute Reviewer's Manual. Joanna Briggs Institute, 2017. Available from: https://wiki.joannabriggs.org/display/MANUAL/JBI+Reviewer%27s+Manual
- 31. Greenacre M. Correspondence analysis in practice. 2nd ed. Boca Raton, FL: Chapman & Hall/CRC 2007.
- 32. Le Roux B, Rouanet H. Multiple correspondence analysis. London: Sage 2010.

- 33. Scheutz F, Heidmann J, Poulsen S. Dental health of Vietnamese boat people on Pulau Bidong, Malaysia. Community Dentistry and Oral Epidemiology. 1983 Aug;11(4):255-8.
- 34. Levy J. Epidemiological survey of intestinal parasitic infections in children of Sabah, Malaysia. Journal of Public Health. 1988 Aug 1;10(3):240-9.
- 35. Kassim MS, George R, Kassim K, Begum M, Cherian MP, Tajudin AK, Chandran V, Anan A, Reddy R, Singh J. Child abuse and neglect as seen in General Hospital, Kuala Lumpur A two year study. The Medical Journal of Malaysia. 1989 Jun;44(2):111-21.
- 36. Zulkifli SN, U KM, Yusof K, Lin WY. Maternal and child health in urban Sabah, Malaysia: a comparison of citizens and migrants. Asia Pacific Journal of Public Health. 1994 Jul;7(3):151-8.
- 37. Rajeswari B, Sinniah B, Hussein H. Socio-economic factors associated with intestinal parasites among children living in Gombak, Malaysia. Asia Pacific Journal of Public Health. 1994 Jan 1:21-5.
- 38. Jeyakumar D. Thiamine responsive ankle oedema in detention centre inmates. The Medical Journal of Malaysia. 1995 Mar;50(1):17-20.
- 39. Jamaiah I, Anuar AK, Najib NA, Zurainee MN. Imported malaria: a retrospective study in University Hospital, Kuala Lumpur, a ten-year experience. Medical Journal of Malaysia. 1998;53(1):6-9.
- 40. Krahl W, Hashim A. Psychiatric disorders in ASEAN-migrants in Malaysia A university hospital experience. The Medical Journal of Malaysia. 1998 Sep;53(3):232-8.
- 41. Zabedah MY, Razak M, Zakiah I, Zuraidah AB. Profile of solvent abusers (glue sniffers) in East Malaysia. The Malaysian Journal of Pathology. 2001 Dec;23(2):105-9.
- 42. Dony JF, Ahmad J, Tiong YK. Epidemiology of tuberculosis and leprosy, Sabah, Malaysia. Tuberculosis. 2004 Jan 1;84(1-2):8-18.
- 43. Chandran PA, Jayaram G, Mahmud R, Anuar AK. Microfilaria in hydrocele fluid cytology. The Malaysian Journal of Pathology. 2004 Dec;26(2):119-23.
- 44. Nissapatorn V, Lim YA, Jamaiah I, Agnes LS, Amyliana K, Wen CC, Nurul H, Nizam S, Quake CT, Valartmathi C, Woei CY. Parasitic infections in Malaysia: Changing and challenges. Southeast Asian Journal of Tropical Medicine and Public Health. 2005;36:50-9.
- 45. Sobri M, Merican JS, Nordiyana M, Valarmathi S, Al-Edrus SA. Neuroimaging features of tuberculous meningitis. Medical Journal of Malaysia. 2006 Mar;61(1):36-40.
- 46. Leong CC. Pre-employment medical examination of Indonesian domestic helpers in a private clinic in Johor Bahru An eight year review. Medical Journal of Malaysia. 2006 Dec;61(5):592.
- 47. Sasidharan S, Uyub AM, Azlan AA. Further evidence of ethnic and gender differences for Helicobacter pylori infection among endoscoped patients. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2008 Dec 1;102(12):1226-32.
- 48. Masitah M. Malaria among foreign workers in Selangor, Malaysia. Journal of Health and Translational Medicine. 2008 Dec 28;11(2):53-8.
- 49. Shailendra S, Prepageran N. Oropharyngeal rhinosporidiosis in a migrant worker: A delayed presentation. Medical Journal of Malaysia. 2008 Mar;63(1):65.

- 50. Chan BT, Amal RN, Noor Hayati MI, Kino H, Anisah N, Norhayati M, Sulaiman O, Mohammed Abdullah M, Fatmah MS, Roslida AR, Ismail G. Seroprevalence of toxoplasmosis among migrant workers from different Asian countries working in Malaysia. Southeast Asian Journal of Tropical Medicine and Public Health. 2008 Jan;39(1):9.
- 51. Farhana F, Jamaiah I, Rohela M, Abdul-Aziz NM, Nissapatorn V. A ten year (1999-2008) retrospective study of amoebiasis in University Malaya Medical Centre (UMMC), Kuala Lumpur, Malaysia. Tropical Biomedicine. 2009 Dec 1;26:262-6.
- 52. Chan BT, Amal RN, Noor Hayati MI, Kino H, Anisah N, Norhayati M. Comparative study of seroprevalence of toxoplasmosis between local workers and migrant workers in Malaysia. Archives of Medical Science. 2009 Apr 1;5(2):255.
- 53. Murty OP. Cystic tumor of papillary muscle of heart: A rare finding in sudden death. The American Journal of Forensic Medicine and Pathology. 2009 Jun 1;30(2):201-3.
- 54. Murty OP, Kian CK, Husin MH, Yusuf WY. Fatal lightning strikes in Malaysia: A review of 27 fatalities. The American Journal of Forensic Medicine and Pathology. 2009 Sep 1;30(3):246-51.
- 55. Mustafa B, Hani AW, Mariam M, Khairul AH, Abdul Rasid K. Epidemiological and clinical features of dengue versus other acute febrile illnesses amongst patients seen at government polyclinics. Medical Journal of Malaysia. 2010 Dec 1;65(4):291-6.
- 56. Su TA, Hoe VC, Masilamani R, Mahmud AB. Hand-arm vibration syndrome among a group of construction workers in Malaysia. Occupational and Environmental Medicine. 2011 Jan 1;68(1):58-63.
- 57. Daher AM, Ibrahim HS, Daher TM, k Anbori A. Health related quality of life among Iraqi immigrants settled in Malaysia. BMC Public Health. 2011 Dec 1;11(1):407.
- 58. Ratnasingam J, Ioras F, Ishak MK. Migrant contract workers and occupational accidents in the furniture industry. Journal of Applied Sciences. 2011 Dec;11(14):2646-51.
- 59. Ab Rahman AK, Abdullah FH. Visceral leishmaniasis (kala-azar) and malaria coinfection in an immigrant in the state of Terengganu, Malaysia: A case report. Journal of Microbiology, Immunology and Infection. 2011 Feb 1;44(1):72-6.
- 60. Taib T, Baba R. Clinical characteristics of patients with leprosy in Hospital Kuala Lumpur. Malaysian Journal of Dermatology. 2013 Jan 1;31:30-6.
- 61. Osman MT, Al-Naggar RA, Taha BI. Knowledge and awareness of cervical cancer screening among Iraqi immigrant women living in Malaysia. World Journal of Medical Sciences. 2013;8(2):123-9.
- 62. Minhat HS, Mustafa J, Sarabi S. Sociodemographic determinants of knowledge on HPV vaccination among Iranian women living in Malaysia. Academic Journal of Cancer Research. 2013;6(1):8-12.
- 63. Mendelsohn JB, Rhodes T, Spiegel P, Schilperoord M, Burton JW, Balasundaram S, Wong C, Ross DA. Bounded agency in humanitarian settings: A qualitative study of adherence to antiretroviral therapy among refugees situated in Kenya and Malaysia. Social Science & Medicine. 2014 Nov 1;120:387-95.
- 64. Mendelsohn JB, Schilperoord M, Spiegel P, Balasundaram S, Radhakrishnan A, Lee CK, Larke N, Grant AD, Sondorp E, Ross DA. Is forced migration a barrier to treatment success? Similar HIV treatment outcomes among refugees and a surrounding host community in Kuala Lumpur, Malaysia. AIDS and Behavior. 2014 Feb 1;18(2):323-34.

- 65. Kwan Z, Pailoor J, Tan LL, Robinson S, Wong SM, Ismail R. Leprosy: An imported disease. Leprosy Review. 2014 Sep 1;85(3):170-6.
- 66. Santos A, Ramos HM, Ramasamy G, Fernandes C. Prevalence of musculoskeletal pain in a sample of migrant workers in the manufacturing industry in Malaysia. Public Health. 2014 Aug;128(8):759.
- 67. Razali S, Salleh RI, Yahya B, Ahmad SH. Maternal filicide among women admitted to forensic psychiatric institutions in Malaysia: Case series. East Asian Archives of Psychiatry. 2015 Jun;25(2):79.
- 68. Elmi OS, Hasan H, Abdullah S, Jeab MZ, Alwi ZB, Naing NN. Multidrug-resistant tuberculosis and risk factors associated with its development: A retrospective study. The Journal of Infection in Developing Countries. 2015 Oct 29;9(10):1076-85.
- 69. Santos A, Ramos HM, Ramasamy G, Fernandes C. Musculoskeletal pain among migrant workers in the Malaysian manufacturing industry: the impact of the physical environment, workload and work patterns. Pertanika Journal of Social Sciences & Humanities. 2015 Jun 1;23:315-24.
- 70. William T, Parameswaran U, Lee WK, Yeo TW, Anstey NM, Ralph AP. Pulmonary tuberculosis in outpatients in Sabah, Malaysia: Advanced disease but low incidence of HIV coinfection. BMC Infectious Diseases. 2015 Dec 1;15(1):32.
- 71. Siah PC, Lee WY, Goh LY. Quality of life among refugee children in Malaysia A brief report. Vulnerable Children and Youth Studies. 2015 Oct 2;10(4):294-9.
- 72. Guinto RL, Curran UZ, Suphanchaimat R, Pocock NS. Universal health coverage in 'One ASEAN': Are migrants included? Global Health Action. 2015 Dec 1;8(1):25749.
- 73. Vijian K, Tata MD, Palayan K. Comparison of features and outcomes of perforated peptic ulcer between Malaysians and foreigners. Medical Journal of Malaysia. 2016 Feb 1;71(1):13.
- 74. Azian MN, Hakim SL, Khadri MS, Yusri MY, Adela JI, Noor M, Nurhainis OS, Karnan G, Shamilah H. Leishmaniasis in Peninsular Malaysia: The role of immigrant workers and the vector. Southeast Asian Journal of Tropical Medicine and Public Health. 2016 Jul 1;47(4):607-16.
- 75. Sahimin N, Lim YA, Ariffin F, Behnke JM, Lewis JW, Zain SN. Migrant workers in Malaysia: Current implications of sociodemographic and environmental characteristics in the transmission of intestinal parasitic infections. PLoS Neglected Tropical Diseases. 2016 Nov;10(11).
- 76. Noh NA, Wahab HA, Bakar Ah SH, Islam MR. Public health services for foreign workers in Malaysia. Social Work in Public Health. 2016 Jul 28;31(5):419-30.
- 77. Kamaludin NA, How V. The inequality of environmental health awareness among foreign immigrants and local blue-collar workers in Kuala Lumpur, Malaysia. Environmental Justice. 2016 Dec 1;9(6):181-7.
- 78. Min NN, Vasudevan SK, Jasman AA, Aisyahbinti A, Myint KT. Work-related ocular injuries in Johor Bahru, Malaysia. International Eye Science. 2016;16(3):416-22.
- 79. Woh PY, Thong KL, Behnke JM, Lewis JW, Zain SN. Characterization of nontyphoidal Salmonella isolates from asymptomatic migrant food handlers in Peninsular Malaysia. Journal of Food Protection. 2017 Aug;80(8):1378-83.

- 80. Tanabe M, Myers A, Bhandari P, Cornier N, Doraiswamy S, Krause S. Family planning in refugee settings: Findings and actions from a multi-country study. Conflict and Health. 2017 Dec;11(1):9.
- 81. Ratnalingam V, Umapathy T, Sumugam K, Hanafi H, Retnasabapathy S. Microbial keratitis in West and East Malaysia. Internal Eye Science. 2017;17(11):1989-92.
- 82. Woh PY, Thong KL, Lim YA, Behnke JM, Lewis JW, Mohd Zain SN. Microorganisms as an indicator of hygiene status among migrant food handlers in Peninsular Malaysia. Asia Pacific Journal of Public Health. 2017 Oct;29(7):599-607.
- 83. Noor NM, Shaker MN. Perceived workplace discrimination, coping and psychological distress among unskilled Indonesian migrant workers in Malaysia. International Journal of Intercultural Relations. 2017 Mar 1;57:19-29.
- 84. Noordin R, Mohd Zain SN, Yunus MH, Sahimin N. Seroprevalence of lymphatic filariasis among migrant workers in Peninsular Malaysia. Transactions of The Royal Society of Tropical Medicine and Hygiene. 2017 Aug 1;111(8):370-2.
- 85. Sahimin N, Lim YA, Ariffin F, Behnke JM, Basáñez MG, Walker M, Lewis JW, Noordin R, Abdullah KA, Zain SN. Socio-demographic determinants of Toxoplasma gondii seroprevalence in migrant workers of Peninsular Malaysia. Parasites & Vectors. 2017 Dec 1;10(1):238.
- 86. Labao HC, Faller EM, Bacayo MF. 'Aches and pains' of Filipino migrant workers in Malaysia: A profile of work-related musculoskeletal disorders. Annals of Global Health. 2018;84(3):474.
- 87. Shaw SA, Ward KP, Pillai V, Hinton DE. A group mental health randomized controlled trial for female refugees in Malaysia. American Journal of Orthopsychiatry. 2018 Jul 23.
- 88. Rahman SA, Jeffree MS, Kamaludin F, Din IA, Yusof M, Ahmed K. Beriberi outbreak among unauthorised immigrants in a detention camp in Malaysia. Journal of Immigrant and Minority Health. 2018 Oct 1;20(5):1294-7.
- 89. Sahimin N, Douadi B, Lim AL, Behnke JM, Zain SN. Distribution of Giardia duodenalis (Assemblages A and B) and Cryptosporidium parvum amongst migrant workers in Peninsular Malaysia. Acta Tropica. 2018 Jun 1;182:178-84.
- 90. Nwabichie CC, Manaf RA, Ismail SB. Factors affecting uptake of cervical cancer screening among African women in Klang Valley, Malaysia. Asian Pacific Journal of Cancer Prevention. 2018;19(3):825.
- 91. Jeffree SM, Ahmed K, Safian N, Hassan R, Mihat O, Lukman KA, Shamsudin SB, Kamaludin F. Falciparum malaria outbreak in Sabah linked to an immigrant rubber tapper. The American Journal of Tropical Medicine and Hygiene. 2018 Jan 10;98(1):45-50.
- 92. Zerguine H, Tamrin SB, Jalaludin J. Prevalence, source and severity of work-related injuries among "foreign" construction workers in a large Malaysian organisation: A cross-sectional study. Industrial Health. 2018:2017-0205.
- 93. Ya'acob NA, Abidin EZ, Rasdi I, Rahman AA, Ismail S. Reducing work-related musculoskeletal symptoms through implementation of Kiken Yochi training intervention approach. Work. 2018 Jan 1;60(1):143-52.
- 94. Loganathan T, Rui D, Ng CW, Pocock NS. Breaking down the barriers: Understanding migrant workers' access to healthcare in Malaysia. PloS one. 2019;14(7).

- 95. Rahman MM, Arif MT, Safii R, Tambi Z, Akoi C, Jantan Z, Halim SA. Care seeking behaviour of Bangladeshi migrant workers in Sarawak, Malaysia. Bangladesh Medical Research Council Bulletin. 2019 Jun 15;45(1):47-53.
- 96. Siah PC, Low SK, Lee WY, Lim YY, Tan JT. Discrimination perceived by child refugees in Malaysia: from the views of representatives from refugees' community service centres and non-government service. Vulnerable Children and Youth Studies. 2019 Sep 1:1-4.
- 97. Sahimin N, Lim YA, Noordin R, Yunus MH, Arifin N, Behnke JM, Zain SN. Epidemiology and immunodiagnostics of Strongyloides stercoralis infections among migrant workers in Malaysia. Asian Pacific Journal of Tropical Medicine. 2019 Jun 1;12(6):250.
- 98. Chuah FL, Tan ST, Yeo J, Legido-Quigley H. Health system responses to the health needs of refugees and asylum-seekers in Malaysia: A qualitative study. International Journal of Environmental Research and Public Health. 2019 Jan;16(9):1584.
- 99. Vearey J, Orcutt M, Gostin L, Braham CA, Duigan P. Building alliances for the global governance of migration and health. BMJ. 2019 Sep 16;366:l4143.
- 100. World Health Organization Regional Office for Europe. Strategy and action plan for refugee and migrant health in the WHO European Region. Copenhagen: World Health Organization Regional Office for Europe 2016.
- 101. Reidpath DD, Allotey P. Infant mortality rate as an indicator of population health. Journal of Epidemiology & Community Health. 2003 May 1;57(5):344-6.
- 102. Vijayasingham L, Rhule E, Asgari-Jirhandeh N, Allotey P. Restrictive migration policies in low-income and middle-income countries. The Lancet Global Health. 2019 Jul 1;7(7):e843-4.
- 103. Juárez SP, Honkaniemi H, Dunlavy AC, Aldridge RW, Barreto ML, Katikireddi SV, Rostila M. Effects of non-health-targeted policies on migrant health: A systematic review and meta-analysis. The Lancet Global Health. 2019 Apr 1;7(4):e420-35.
- 104. Shao C, Meng X, Cui S, Wang J, Li C. Income-related health inequality of migrant workers in China and its decomposition: An analysis based on the 2012 China labor-force dynamics survey data. Journal of the Chinese Medical Association. 2016 Oct 1;79(10):531-7.
- 105. He X, Wong DF. A comparison of female migrant workers' mental health in four cities in China. International Journal of Social Psychiatry. 2013 Mar;59(2):114-22.
- 106. Al-Khatib IA, Ju'ba A, Kamal N, Hamed N, Hmeidan N, Massad S. Impact of housing conditions on the health of the people at al-Ama'ri refugee camp in the West Bank of Palestine. International Journal of Environmental Health Research. 2003 Dec 1;13(4):315-26.
- 107. Mutalip MH, Mishkin K, Paiwai F, Sulaiman J, Yoep N. Factors associated with sexual intercourse, condom-use, and perceived peer behaviors among adolescents in Malaysia: A school-based cross-sectional study. Malaysian Journal of Social Sciences & Humanities. 2019 May 1;4(2):8-18.
- 108. Khan MA, Rosly NA, Majeed AB, Ismail NE. Knowledge of modes of HIV transmission among public secondary school students in urban and rural areas in Malaysia. Pakistan Journal of Pharmaceutical Sciences. 2018 Sep 1;31(5).
- 109. Shakir M, Mahavera S, Wong LP, Lim Abdullah K, Adam P. Online STI information seeking behaviour and condom use intentions among young Facebook users in Malaysia. Health Promotion International. 2019.

- 110. Renzaho AM, Burns C. Post-migration food habits of sub-Saharan African migrants in Victoria: A cross-sectional study. Nutrition & Dietetics. 2006 Jun;63(2):91-102.
- 111. Ganle JK, Amoako D, Baatiema L, Ibrahim M. Risky sexual behaviour and contraceptive use in contexts of displacement: Insights from a cross-sectional survey of female adolescent refugees in Ghana. International Journal for Equity in Health. 2019 Dec 1;18(1):127.
- 112. Bosdriesz JR, Lichthart N, Witvliet MI, Busschers WB, Stronks K, Kunst AE. Smoking prevalence among migrants in the US compared to the US-born and the population in countries of origin. PloS One. 2013;8(3).
- 113. Ritchie H, Roser M. Smoking. Oxford: Our World Data; 2019 [updated 2019 Nov; cited 2019 Nov 20]. Available from: https://ourworldindata.org/smoking
- 114. Sweileh WM. Global output of research on the health of international migrant workers from 2000 to 2017. Globalization and Health. 2018 Dec 1;14(1):105.
- 115. World Health Organisation. Noncommunicable diseases (NCD) country profiles Malaysia. (n.p.): World Health Organisation 2018.
- 116. Zimmerman C, Hossain M, Watts C. Human trafficking and health: A conceptual model to inform policy, intervention and research. Social Science & Medicine. 2011 Jul 1;73(2):327-35.
- 117. Zimmerman C, Kiss L, Hossain M. Migration and health: A framework for 21st century policy-making. PloS Medicine. 2011 May;8(5).
- 118. Par M, Hassan SA, Uba I, Baba M. Perceived stress among International postgraduate students in Malaysia. International Journal of Psychological Studies. 2015 Dec 1;7(4):1.
- 119. Kohno A, Farid ND, Musa G, Aziz NA, Nakayama T, Dahlui M. Factors affecting Japanese retirees' healthcare service utilisation in Malaysia: A qualitative study. BMJ Open. 2016 Mar 1;6(3).
- 120. Asits S, Noorhidayah I, Osman A. Migrasi bermusim: Pengaruhnya terhadap kawalan malaria di Sabah. Medical Journal of Malaysia. 1999;54(2):200-209.
- 121. World Health Organization. A conceptual framework for action on the social determinants of health. Geneva: World Health Organization 2010.

VERSION 2 - REVIEW

REVIEWER	Daniel Vujcich
	Curtin University, Australia
REVIEW RETURNED	12-Nov-2020

GENERAL COMMENTS	Thank you for addressing our feedback. I am generally satisfied with the revisions.
	In my view, two points still need to be addressed:
	(1) point 21 - I am not seeking an explanation for the cut-off year, rather the paper requires a statement of the RELEVANCE of such a broad retrospective review. Basically, as a reader, I want to know why a review going back to 1965 is needed and valuable (as opposed to a more focused, in-depth review of more contemporary literature).

(2) point 27 - the limitation section states "only English language articles were included, resulting in the exclusion of one paper in Bahasa Malaysia". I think this statement could create a misleading impression. One reason that you did not find more relevant articles in Bahasa Malaysia may be the fact that you did not include Malaysian keywords in your search terms and you did not specifically search Malaysian language journals and databases. Certainly, language biases are common in literature reviews; however, I feel that the limitation is particularly pertinent in this case because you are seeking to provide a comprehensive review of the literature from/about one country without devising a search strategy that is tailored to the official language of that country. I need more information to be convinced that this is not a serious problem with your study design.

The following reference may be useful:

https://systematicreviewsjournal.biomedcentral.com/articles/10.1186/s13643-018-0786-6.

The Cochrane Handbook also notes that "Removing language restrictions in English language databases is not a good substitute for searching non-English language journals and databases":

https://training.cochrane.org/handbook/current/chapter-04

REVIEWER	Gayle Halas University of Manitoba. Canada
REVIEW RETURNED	15-Nov-2020

GENERAL COMMENTS	For authors:
	Line 24: needs an additional word - "Procuring proper health
	care"
	Line 32: could "self-developed" be changed to something like "a
	decision tree developed by the project team" or for simplicity in the
	abstract: ':a newly developed decision tree"
	Line 38 some wording lending confusion. Does the following
	change convey the intended message?
	On average, the quality of the papers was low, yet quality differed
	significantly among papers using various research designs and
	addressing the range of health dimensions.
	Line 105: Related to use of the word "evidence"more accurate
	to state "evidence on migrant health <concerns issues="" or="" or<="" th=""></concerns>
	needs>"
	Line 126: JBI needs a citation
	Line 128: more clarity on what those variables are?BARHII
	dimensions or factors and levels of evidence?
	Line 120-132: This addition doesn't really speak to patient and
	public involvement since the presentation was delivered to
	stakeholders working on migrant health. I would suggest removing
	so as not to confuse the puprpose of engaging individuals with
	lived experience. Second, simply saying a presentation was
	provided without conveying their feedback and how it was relevant
	to this work, is not useful.
	Line 147: change to "which was" (singular referring to
	population)
	Line 180: clarity neededplease identify where these additional
	records came from. Also, not clear on why it is explicitly stated that
	records published before 1965 are included – this runs contrary to
	what is stated in the exclusion criteria re including Malaysia and
	not Singapore.
	206 studies without Singapore;
	Line 187: "looking over" too parochial for a systematic review
	Line 221: should say "inclusion criteria"

Line 230: This is the first mention of "subdimensions" and although described later and in Figure 4, it would be beneficial to further describe at this point in the paper.

Line 232: still a weakness. I think the authors could provide some argument about the reliability of the data extraction process, based on the level of interpretation that was needed (or not needed) to extract the data. For example, if fairly objective indicators (such as cancer diagnosis) were extracted into the disease and injury category, then one would expect a more trustworthy process of data extraction.

Line 244: not clear on what is being adapted.

Lines 277-279: If the modified checklists are not provided, inserting an example of a modification would be helpful. Results section:

Table 4 and results sections should be organized similarly (ie following BARHII health dimensions).

Line 364-366: Interesting statement but in my opinion, how does it relate to your review question? That needs to be explicit.

Line 399: Spelling: Injury

Line 488: no research gaps are presented in the findings. Not clear where Table 7 recommendations are coming from.

Lines 611-627 – this minimizes your entire study! Suggesting an alternate approach (or alternate tools) suggests an entirely different study. I don't believe what is stated here is a limitation, however maybe just one or two lines that simply suggests the use of additional elements (such as...X.Y.Z...) would be beneficial for future study.

Figure 1 – permission to use this graphic?

VERSION 2 – AUTHOR RESPONSE

Response to Reviewer 3 Comments

Point 3: I am not seeking an explanation for the cut-off year, rather the paper requires a statement of the RELEVANCE of such a broad retrospective review. Basically, as a reader, I want to know why a review going back to 1965 is needed and valuable (as opposed to a more focused, in-depth review of more contemporary literature).

Response 3: We have included papers from 1965 onwards to identify if specific trends were happening (e.g., did the specific focus area in migrant health research change over the years; did the quality of the papers improve over the years?) As a result, we learned that the data set was quite heterogenous, and, hence, did not observe any specific trends in the research area (within migrant health) or in the quality of the papers. We have added this statement to the paper.

Page 7, lines 107-110

Therefore, this study aims to map the existing academic literature on migrant health in Malaysia since 1965 to identify the trends and gaps in this field, as well as to present an overview of the topical coverage, quality, and level of evidence of these scientific studies

The literature was first assessed to understand the trends and topical coverage of research against the six dimensions of the BARHII public health framework.

Page 48, lines 514-517

In terms of research trends, no specific changes in health dimension or quality have been observed over the last six decades. However, it should be noted that qualitative research made its entry in the early 2010s and made up a vast amount of the papers published in recent years.

Point 4: the limitation section states "only English language articles were included, resulting in the exclusion of one paper in Bahasa Malaysia". I think this statement could create a misleading impression. One reason that you did not find more relevant articles in Bahasa Malaysia may be the fact that you did not include Malaysian keywords in your search terms and you did not specifically search Malaysian language journals and databases. Certainly, language biases are common in literature reviews; however, I feel that the limitation is particularly pertinent in this case because you are seeking to provide a comprehensive review of the literature from/about one country without devising a search strategy that is tailored to the official language of that country. I need more information to be convinced that this is not a serious problem with your study design.

Response 4: Thank you for the thorough explanation, and we agree with your point. We have modified this section.

Page 53, lines 627-632

Also, only English language articles were included, resulting in the exclusion of one identified paper in Bahasa Malaysia (the Malay language). Aggravating the issue, other Malaysian articles might not have been identified due to the lack of Malaysian keywords in the search strategy. As a result, much relevant information that could potentially be used to inform both policies and practice, as well as to make this review more comprehensive, may have been excluded from this review.

Response to Reviewer 4 Comments

Point 5: Line 24: needs an additional word - "Procuring proper health care..."

Response 5: We have modified this sentence.

Page 2, lines 23-24

Background: A large number of international migrants in Malaysia face challenges in obtaining good health, the extent of which is still relatively unknown.

Point 6: Line 32: could "self-developed" be changed to something like "a decision tree developed by the project team" or for simplicity in the abstract: ':...a newly developed decision tree..."

Response 6: We have modified this sentence.

Page 2, lines 31-33

Data were extracted by using the BARHII framework and a newly developed decision tree to identify the type of study design and corresponding level of evidence.

Point 7: Line 38 some wording lending confusion. Does the following change convey the intended message?

"On average, the quality of the papers was low, yet quality differed significantly among papers using various research designs and addressing the range of health dimensions."

Response 7: We have modified this sentence.

Page 2, lines 38-39

The average quality of the papers was low, yet quality differed significantly among them.

Point 8: Line 105: Related to use of the word "evidence"....more accurate to state "evidence on migrant health <concerns or issues or needs>..."

Response 8: We have modified this sentence.

Pages 6-7, lines 100-102

Scoping reviews can be helpful to map the academic literature and have been used by different researchers to present the available evidence on migrant health issues in other countries.¹⁵

Point 9: Line 126: JBI needs a citation

Response 9: We have added the reference.

Page 8, lines 122-123

Subsequently, a quality assessment of the included literature was conducted by using the Joanna Briggs Institute (JBI) critical appraisal toolkit.¹⁹

Point 10: Line 128: more clarity on what those variables are? ...BARHII dimensions or factors and levels of evidence?

Response 10: We have modified this sentence.

Page 8, lines 123-126

Lastly, the data was analysed, and a multiple-correspondence analysis (MCA) was applied to explore existing relationships between variables, including the type of migrant, main health dimension, quality of the study, and research design.

Point 11: Line 130-132: This addition doesn't really speak to patient and public involvement since the presentation was delivered to stakeholders working on migrant health. I would suggest removing so as not to confuse the purpose of engaging individuals with lived experience. Second, simply saying a presentation was provided without conveying their feedback and how it was relevant to this work, is not useful.

Response 11: We have removed this part from the patient and public involvement statement.

Page 8, line 128

Patients and the public were not involved in this study.

Point 12: Line 147: change to "which was..." (singular referring to population)

Response 12: We have modified this sentence.

Page 9, lines 143-145

The social inequities element was incorporated by describing the population of interest, which was divided into three categories: foreign workers, asylum seekers and refugees, and unclassified migrants.

Point 13: Line 180: clarity needed...please identify where these additional records came from. Also, not clear on why it is explicitly stated that records published before 1965 are included – this runs contrary to what is stated in the exclusion criteria re including Malaysia and not Singapore.

Response 13: We have modified this sentence.

Page 10, lines 177-179

However, all identified records were retrieved from the six databases to manually screen the data for publication date-related issues.

Point 14: Line 187: "looking over" too parochial for a systematic review

Response 14: We have changed the verb.

Page 11, lines 181-185

The first stage focused on identifying English-language key words and Medical Subject Headings (MeSH) terms for migrants (e.g., immigrants, foreign workers, refugees), health (e.g., disease, infection, disorder), and Malaysia (e.g., Sabah, Kuala Lumpur) through reading search strategies of other review studies on migrant health as well as utilising medical terminology of renowned medical institutions, such as the Mayo Clinic.

Point 15: Line 221: should say "inclusion criteria"

Response 15: We have changed "selection criteria" to "inclusion criteria"

Page 12, lines 217-218

The first stage involved screening titles and abstracts according to the inclusion criteria.

Point 16: Line 230: This is the first mention of "subdimensions" and although described later and in Figure 4, it would be beneficial to further describe at this point in the paper.

Response 16: We have introduced the subdimensions in the 'Conceptual framework' section.

Page 9, lines 140-141

In addition, each health dimension contains various subdimensions (as presented in Figure 1).

Point 17: Line 232: still a weakness. I think the authors could provide some argument about the reliability of the data extraction process, based on the level of interpretation that was needed (or not needed) to extract the data. For example, if fairly objective indicators (such as cancer diagnosis) were extracted into the disease and injury category, then one would expect a more trustworthy process of data extraction.

Response 17: As point 17 and 24 are addressing similar issues, we present a thorough explanation (that includes this point) in point 24. Please, see point 24 for more information.

Point 18: Line 244: not clear on what is being adapted.

Response 18: We have reworded this sentence.

Page 13, lines 238-240

First, a table was created that included definitions of various research designs, and, subsequently, specific traits of these definitions were used to develop guiding questions for the decision tree.

Point 19: Line 277-279: If the modified checklists are not provided, inserting an example of a modification would be helpful.

Response 19: We have added an example. Please see Table 4 on page 16.

Page 15, lines 276-277

An example is provided in Table 4.

Point 20: Table 4 and results sections should be organized similarly (ie following BARHII health

dimensions).

Response 20: Thank you for your suggestion. We decided to keep this structure, as the references

are presented in chronological order.

Point 21: Line 364-366: Interesting statement but in my opinion, how does it relate to your review

question? That needs to be explicit.

Response 21: This information contributes to addressing one of the research objectives – creating a

comprehensive overview of migrant health in Malaysia – by providing a description of the included study. However, we realised that this was not clearly stated in the 'Introduction' (research objectives

paragraph), and, therefore, have modified this section.

Page 7, lines 107-110.

Therefore, this study aims to map the existing academic literature on migrant health in Malaysia since 1965 to identify the trends and gaps in this field, as well as to present an

overview of the topical coverage, quality, and level of evidence of these scientific studies.

Point 22: Line 399: Spelling: Injury

Response 22: We have corrected the word

Page 40, line 414

Disease and injury

Point 23: Line 488: no research gaps are presented in the findings. Not clear where Table 7

recommendations are coming from.

50

Response 23: Thank you for this comment. We have revised this section thoroughly.

Page 48, lines 505-517

The majority of these studies focus on the 'disease and injury' dimension, especially infectious diseases, and includes mostly foreign workers. Two health dimensions (institutional inequality, and morbidity and mortality) as well as various subdimensions of each health dimension, are lacking substantial research. In addition, only a few papers include the asylum seeker and refugee population, and a vast amount do not provide any details to classify the type of migrant. The average quality of the papers was low, yet quality differed significantly among the studies. High-quality studies were mostly qualitative designs that included refugees and focused on living conditions, while prevalence and analytical cross-sectional studies were mostly low quality. In terms of research trends, no specific changes in type of migrant, health dimension or quality of the study have been observed over the last six decades. However, it should be noted that qualitative research made its entry in the early 2010s and made up a vast amount of the papers published in recent years.

Point 24: Lines 611-627 – this minimizes your entire study! Suggesting an alternate approach (or alternate tools) suggests an entirely different study. I don't believe what is stated here is a limitation, however maybe just one or two lines that simply suggests the use of additional elements (such as...X.Y.Z...) would be beneficial for future study.

Response 24: Thank you for the feedback regarding this matter. We have revised this section thoroughly. We have removed the following sentence from the 'Data extraction' section (previously, page 13, lines 231-232):

Data extraction and categorisation into the BARHII framework categories was not cross-checked by a second reviewer due to time and human resource constraints.

Subsequently, we have addressed this matter in the 'Limitations' section.

Pages 53-54, lines 633-651

Inter-rater reliability was limited to a 20% sample of the records in the first (abstract and title) screening stage, and no data extraction nor quality assessment was verified by a second reviewer due to time and resource constraints. Yet, we anticipate low selection bias as the health dimensions in the BARHII framework present clear distinctions between each other.

and most of the included papers used objective indictors. For example, when a paper was measuring the knowledge and awareness regarding the pap smear test among female migrants, it would be classified as a 'risk behaviour' study. Furthermore, we believe that the development of the decision tree and additional objective criteria for the JBI tools – an example was given earlier in Table 4 – reduced the subjectivity of this study, and, hence, increases the reliability. Yet, future research is needed to validate both the decision tree and modified JBI toolkit.

Besides the BARHII framework, various conceptual public health models are available, and these models may include different (sub)dimensions. For instance, the WHO Commission on Social Determinants of Health (CSDH) framework includes material circumstances, such as food availability, whereas this dimension is not included in the BARHII framework. ¹²¹ Similarly, critical appraisal tools other than the JBI toolkit are available, which could address different points to determine the quality of the study. Therefore, it would be helpful to assess other public health models and critical appraisal tools to see if they include additional elements (e.g., food availability) that would be beneficial for future studies.

Point 25: Figure 1 – permission to use this graphic?

Response 25: Yes, we have sought permission from BARHII. Please see the 'Acknowledgement' and 'Consent for publication' sections.

Page 56, lines 678-681

Acknowledgements:

We would like to thank Andrew Seko for granting us permission on behalf of the Bay Area Regional Health Inequities Initiative (BARHII) for the use of their public health framework in this scoping review.

Page 56, lines 687-688

Consent for publication:

Consent regarding the use of the BARHII framework was granted by BARHII.

References

- 1. International Organization for Migration. World migration report 2020. Geneva: International Organization for Migration 2019.
- 2. International Organization for Migration. Malaysia [Internet]. Kuala Lumpur: International Organization for Migration; 2020 [cited 2020 Oct 01]. Available from: https://www.iom.int/countries/malaysia
- 3. Department of Statistics Malaysia. Population quick info [Internet]. Putrajaya: Department of Statistics Malaysia; 2020 [updated 2020 Apr 30; cited 2020 Apr 30]. Available from: https://www.dosm.gov.my/v1/index.php?r=column/cone&menu_id=b2RVUmEyZ2ZZYXhaUzUrdnpsa Wozdz09
- 4. Department of Statistics Malaysia. Current population estimates Malaysia 2019. Putrajaya: Department of Statistics Malaysia.
- 5. Office of the United Nations High Commissioner for Human Rights. The rights of non-citizens. Geneva: United Nations 2006.
- 6. International Organization for Migration. Glossary on migration. 2nd ed. Geneva: International Organization for Migration 2011.
- 7. Ministry of Home Affairs Malaysia. Statistik pekerja asing terkini mengikut negeri dan sektor [Internet]. Putrajaya: Ministry of Home Affairs Malaysia; 2019 [updated 2019 Aug 1; cited 2020 Apr 14]. Available from: http://www.data.gov.my/data/ms_MY/dataset/statistik-pekerja-asing-terkini-mengikut-negeri-dan-sektor
- 8. United Nations High Commissioner for Refugees. Figures at a glance in Malaysia [Internet]. Kuala Lumpur: United Nations High Commissioner for Refugees; 2019 [cited 2019 Nov 20]. Available from: https://www.unhcr.org/en-my/figures-at-a-glance-in-malaysia.html
- 9. Chuah FL, Tan ST, Yeo J, Legido-Quigley H. The health needs and access barriers among refugees and asylum-seekers in Malaysia: a qualitative study. International Journal for Equity in Health. 2018 Dec 1;17(1):120.
- 10. Devadason ES, Meng CW. Policies and laws regulating migrant workers in Malaysia: A critical appraisal. Journal of Contemporary Asia. 2014 Jan 2;44(1):19-35.
- 11. International Organization for Migration. IOM handbook: protection and assistance for migrants vulnerable to violence, exploitation and abuse. Geneva: International Organization for Migration 2018.
- 12. Pocock NS, Suphanchaimat R, Chan CK, Faller EM, Harrigan N, Pillai V, Wickramage K. Reflections on migrant and refugee health in Malaysia and the ASEAN region. BMC Proceedings. 2018 Apr;12(Suppl 4):4.
- 13. Lasimbang HB, Tong WT, Low WY. Migrant workers in Sabah, East Malaysia: The importance of legislation and policy to uphold equity on sexual and reproductive health and rights. Best Practice & Research Clinical Obstetrics & Gynaecology. 2016 Apr 1;32:113-23.
- 14. Sweileh WM, Wickramage K, Pottie K, Hui C, Roberts B, Sawalha AF, Zyoud SH. Bibliometric analysis of global migration health research in peer-reviewed literature (2000–2016). BMC Public Health. 2018 Dec;18(1):777.
- 15. Villarroel N, Hannigan A, Severoni S, Puthoopparambil S, MacFarlane A. Migrant health research in the Republic of Ireland: a scoping review. BMC Public Health. 2019 Dec;19(1):324.

- 16. Khanlou N, Haque N, Skinner A, Mantini A, Kurtz Landy C. Scoping review on maternal health among immigrant and refugee women in Canada: prenatal, intrapartum, and postnatal care. Journal of Pregnancy 2017 Jan;2017:1-14
- 17. Tricco AC, Lillie E, Zarin W, O'Brien KK, Colquhoun H, Levac D, Moher D, Peters MD, Horsley T, Weeks L, Hempel S. PRISMA extension for scoping reviews (PRISMA-ScR): Checklist and explanation. Annals of Internal Medicine. 2018 Oct 2;169(7):467-73.
- 18. Bay Area Regional Health Inequities Initiative. Framework [Internet]. Oakland, CA: Bay Area Regional Health Inequities Initiative; 2015 [cited 2019 Sep 18]. Available from: http://barhii.org/framework/
- 19. Aromataris E, Munn Z, editors. *Joanna Briggs Institute Reviewer's Manual*. Joanna Briggs Institute, 2017. Available from: https://wiki.joannabriggs.org/display/MANUAL/JBI+Reviewer%27s+Manual
- 20. National Academies of Sciences, Engineering, and Medicine. A framework for educating health professionals to address the social determinants of health. Washington, DC: The National Academies Press 2016.
- 21. Public Health England. Public Health Outcomes Framework 2019/20: a consultation. London: Public Health England 2019.
- 22. London School of Hygiene & Tropical Medicine. Which databases should I search? A guide to the most popular resources. London: London School of Hygiene & Tropical Medicine 2014.
- 23. Bramer WM, Rethlefsen ML, Kleijnen J, Franco OH. Optimal database combinations for literature searches in systematic reviews: a prospective exploratory study. Systematic Reviews. 2017 Dec 1;6(1):245.
- 24. Centre for Evidence-Based Medicine. Study designs. Oxford: Centre for Evidence-Based Medicine; 2019 [cited 2019 Oct 10]. Available from: https://www.cebm.net/2014/04/study-designs/
- 25. Centre for Cognitive Ageing and Cognitive Epidemiology. Systematic reviews and metaanalyses: A step-by-step guide. Edinburgh: Centre for Cognitive Ageing and Cognitive Epidemiology; 2013 [cited 2019 Oct 15]. Available from: https://www.ccace.ed.ac.uk/research/softwareresources/systematic-reviews-and-meta-analyses
- 26. Alexander LK, Lopes B, Ricchetti-Masterson K, Yeatts KB. ERIC notebook [2nd ed]. Chapel Hill: Epidemiologic Research and Information Center, 2015.
- 27. Cochrane Community. Glossary. London: Cochrane Community; 2019 [cited 2019 Nov 2]. Available from: https://community.cochrane.org/glossary#letter-N
- 28. Peacock JL, Peacock PJ. Oxford Handbook of Medical Statistics. Oxford: Oxford University Press 2011.
- 29. Tomlin G, Borgetto B. Research pyramid: A new evidence-based practice model for occupational therapy. American Journal of Occupational Therapy. 2011 Mar 1;65(2):189-96.
- 30. Moola S, Munn Z, Tufanaru C, Aromataris E, Sears K, Sfetcu R, Currie M, Lisy K, Qureshi R, Mattis P, Mu P. Chapter 7: Systematic reviews of etiology and risk. In: Aromataris E, Munn Z (Editors). *JBI Manual for Evidence Synthesis*. JBI, 2017. Available from: https://doi.org/10.46658/JBIMES-20-08

- 31. Greenacre M. Correspondence analysis in practice. 2nd ed. Boca Raton, FL: Chapman & Hall/CRC 2007.
- 32. Le Roux B, Rouanet H. Multiple correspondence analysis. London: Sage 2010.
- 33. Scheutz F, Heidmann J, Poulsen S. Dental health of Vietnamese boat people on Pulau Bidong, Malaysia. Community Dentistry and Oral Epidemiology. 1983 Aug;11(4):255-8.
- 34. Levy J. Epidemiological survey of intestinal parasitic infections in children of Sabah, Malaysia. Journal of Public Health. 1988 Aug 1;10(3):240-9.
- 35. Kassim MS, George R, Kassim K, Begum M, Cherian MP, Tajudin AK, Chandran V, Anan A, Reddy R, Singh J. Child abuse and neglect as seen in General Hospital, Kuala Lumpur A two year study. The Medical Journal of Malaysia. 1989 Jun;44(2):111-21.
- 36. Zulkifli SN, U KM, Yusof K, Lin WY. Maternal and child health in urban Sabah, Malaysia: a comparison of citizens and migrants. Asia Pacific Journal of Public Health. 1994 Jul;7(3):151-8.
- 37. Rajeswari B, Sinniah B, Hussein H. Socio-economic factors associated with intestinal parasites among children living in Gombak, Malaysia. Asia Pacific Journal of Public Health. 1994 Jan 1:21-5.
- 38. Jeyakumar D. Thiamine responsive ankle oedema in detention centre inmates. The Medical Journal of Malaysia. 1995 Mar;50(1):17-20.
- 39. Jamaiah I, Anuar AK, Najib NA, Zurainee MN. Imported malaria: a retrospective study in University Hospital, Kuala Lumpur, a ten-year experience. Medical Journal of Malaysia. 1998;53(1):6-9.
- 40. Krahl W, Hashim A. Psychiatric disorders in ASEAN-migrants in Malaysia A university hospital experience. The Medical Journal of Malaysia. 1998 Sep;53(3):232-8.
- 41. Zabedah MY, Razak M, Zakiah I, Zuraidah AB. Profile of solvent abusers (glue sniffers) in East Malaysia. The Malaysian Journal of Pathology. 2001 Dec;23(2):105-9.
- 42. Dony JF, Ahmad J, Tiong YK. Epidemiology of tuberculosis and leprosy, Sabah, Malaysia. Tuberculosis. 2004 Jan 1;84(1-2):8-18.
- 43. Chandran PA, Jayaram G, Mahmud R, Anuar AK. Microfilaria in hydrocele fluid cytology. The Malaysian Journal of Pathology. 2004 Dec;26(2):119-23.
- 44. Nissapatorn V, Lim YA, Jamaiah I, Agnes LS, Amyliana K, Wen CC, Nurul H, Nizam S, Quake CT, Valartmathi C, Woei CY. Parasitic infections in Malaysia: Changing and challenges. Southeast Asian Journal of Tropical Medicine and Public Health. 2005;36:50-9.
- 45. Sobri M, Merican JS, Nordiyana M, Valarmathi S, Al-Edrus SA. Neuroimaging features of tuberculous meningitis. Medical Journal of Malaysia. 2006 Mar;61(1):36-40.
- 46. Leong CC. Pre-employment medical examination of Indonesian domestic helpers in a private clinic in Johor Bahru An eight year review. Medical Journal of Malaysia. 2006 Dec;61(5):592.
- 47. Sasidharan S, Uyub AM, Azlan AA. Further evidence of ethnic and gender differences for Helicobacter pylori infection among endoscoped patients. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2008 Dec 1;102(12):1226-32.
- 48. Masitah M. Malaria among foreign workers in Selangor, Malaysia. Journal of Health and Translational Medicine. 2008 Dec 28;11(2):53-8.

- 49. Shailendra S, Prepageran N. Oropharyngeal rhinosporidiosis in a migrant worker: A delayed presentation. Medical Journal of Malaysia. 2008 Mar;63(1):65.
- 50. Chan BT, Amal RN, Noor Hayati MI, Kino H, Anisah N, Norhayati M, Sulaiman O, Mohammed Abdullah M, Fatmah MS, Roslida AR, Ismail G. Seroprevalence of toxoplasmosis among migrant workers from different Asian countries working in Malaysia. Southeast Asian Journal of Tropical Medicine and Public Health. 2008 Jan;39(1):9.
- 51. Farhana F, Jamaiah I, Rohela M, Abdul-Aziz NM, Nissapatorn V. A ten year (1999-2008) retrospective study of amoebiasis in University Malaya Medical Centre (UMMC), Kuala Lumpur, Malaysia. Tropical Biomedicine. 2009 Dec 1;26:262-6.
- 52. Chan BT, Amal RN, Noor Hayati MI, Kino H, Anisah N, Norhayati M. Comparative study of seroprevalence of toxoplasmosis between local workers and migrant workers in Malaysia. Archives of Medical Science. 2009 Apr 1;5(2):255.
- 53. Murty OP. Cystic tumor of papillary muscle of heart: A rare finding in sudden death. The American Journal of Forensic Medicine and Pathology. 2009 Jun 1;30(2):201-3.
- 54. Murty OP, Kian CK, Husin MH, Yusuf WY. Fatal lightning strikes in Malaysia: A review of 27 fatalities. The American Journal of Forensic Medicine and Pathology. 2009 Sep 1;30(3):246-51.
- 55. Mustafa B, Hani AW, Mariam M, Khairul AH, Abdul Rasid K. Epidemiological and clinical features of dengue versus other acute febrile illnesses amongst patients seen at government polyclinics. Medical Journal of Malaysia. 2010 Dec 1;65(4):291-6.
- 56. Su TA, Hoe VC, Masilamani R, Mahmud AB. Hand-arm vibration syndrome among a group of construction workers in Malaysia. Occupational and Environmental Medicine. 2011 Jan 1;68(1):58-63.
- 57. Daher AM, Ibrahim HS, Daher TM, k Anbori A. Health related quality of life among Iraqi immigrants settled in Malaysia. BMC Public Health. 2011 Dec 1;11(1):407.
- 58. Ratnasingam J, Ioras F, Ishak MK. Migrant contract workers and occupational accidents in the furniture industry. Journal of Applied Sciences. 2011 Dec;11(14):2646-51.
- 59. Ab Rahman AK, Abdullah FH. Visceral leishmaniasis (kala-azar) and malaria coinfection in an immigrant in the state of Terengganu, Malaysia: A case report. Journal of Microbiology, Immunology and Infection. 2011 Feb 1;44(1):72-6.
- 60. Taib T, Baba R. Clinical characteristics of patients with leprosy in Hospital Kuala Lumpur. Malaysian Journal of Dermatology. 2013 Jan 1;31:30-6.
- 61. Osman MT, Al-Naggar RA, Taha BI. Knowledge and awareness of cervical cancer screening among Iraqi immigrant women living in Malaysia. World Journal of Medical Sciences. 2013;8(2):123-9.
- 62. Minhat HS, Mustafa J, Sarabi S. Sociodemographic determinants of knowledge on HPV vaccination among Iranian women living in Malaysia. Academic Journal of Cancer Research. 2013;6(1):8-12.
- 63. Mendelsohn JB, Rhodes T, Spiegel P, Schilperoord M, Burton JW, Balasundaram S, Wong C, Ross DA. Bounded agency in humanitarian settings: A qualitative study of adherence to antiretroviral therapy among refugees situated in Kenya and Malaysia. Social Science & Medicine. 2014 Nov 1;120:387-95.

- 64. Mendelsohn JB, Schilperoord M, Spiegel P, Balasundaram S, Radhakrishnan A, Lee CK, Larke N, Grant AD, Sondorp E, Ross DA. Is forced migration a barrier to treatment success? Similar HIV treatment outcomes among refugees and a surrounding host community in Kuala Lumpur, Malaysia. AIDS and Behavior. 2014 Feb 1;18(2):323-34.
- 65. Kwan Z, Pailoor J, Tan LL, Robinson S, Wong SM, Ismail R. Leprosy: An imported disease. Leprosy Review. 2014 Sep 1;85(3):170-6.
- 66. Santos A, Ramos HM, Ramasamy G, Fernandes C. Prevalence of musculoskeletal pain in a sample of migrant workers in the manufacturing industry in Malaysia. Public Health. 2014 Aug;128(8):759.
- 67. Razali S, Salleh RI, Yahya B, Ahmad SH. Maternal filicide among women admitted to forensic psychiatric institutions in Malaysia: Case series. East Asian Archives of Psychiatry. 2015 Jun;25(2):79.
- 68. Elmi OS, Hasan H, Abdullah S, Jeab MZ, Alwi ZB, Naing NN. Multidrug-resistant tuberculosis and risk factors associated with its development: A retrospective study. The Journal of Infection in Developing Countries. 2015 Oct 29;9(10):1076-85.
- 69. Santos A, Ramos HM, Ramasamy G, Fernandes C. Musculoskeletal pain among migrant workers in the Malaysian manufacturing industry: the impact of the physical environment, workload and work patterns. Pertanika Journal of Social Sciences & Humanities. 2015 Jun 1;23:315-24.
- 70. William T, Parameswaran U, Lee WK, Yeo TW, Anstey NM, Ralph AP. Pulmonary tuberculosis in outpatients in Sabah, Malaysia: Advanced disease but low incidence of HIV coinfection. BMC Infectious Diseases. 2015 Dec 1;15(1):32.
- 71. Siah PC, Lee WY, Goh LY. Quality of life among refugee children in Malaysia A brief report. Vulnerable Children and Youth Studies. 2015 Oct 2;10(4):294-9.
- 72. Guinto RL, Curran UZ, Suphanchaimat R, Pocock NS. Universal health coverage in 'One ASEAN': Are migrants included? Global Health Action. 2015 Dec 1;8(1):25749.
- 73. Vijian K, Tata MD, Palayan K. Comparison of features and outcomes of perforated peptic ulcer between Malaysians and foreigners. Medical Journal of Malaysia. 2016 Feb 1;71(1):13.
- 74. Azian MN, Hakim SL, Khadri MS, Yusri MY, Adela JI, Noor M, Nurhainis OS, Karnan G, Shamilah H. Leishmaniasis in Peninsular Malaysia: The role of immigrant workers and the vector. Southeast Asian Journal of Tropical Medicine and Public Health. 2016 Jul 1;47(4):607-16.
- 75. Sahimin N, Lim YA, Ariffin F, Behnke JM, Lewis JW, Zain SN. Migrant workers in Malaysia: Current implications of sociodemographic and environmental characteristics in the transmission of intestinal parasitic infections. PLoS Neglected Tropical Diseases. 2016 Nov;10(11).
- 76. Noh NA, Wahab HA, Bakar Ah SH, Islam MR. Public health services for foreign workers in Malaysia. Social Work in Public Health. 2016 Jul 28;31(5):419-30.
- 77. Kamaludin NA, How V. The inequality of environmental health awareness among foreign immigrants and local blue-collar workers in Kuala Lumpur, Malaysia. Environmental Justice. 2016 Dec 1;9(6):181-7.
- 78. Min NN, Vasudevan SK, Jasman AA, Aisyahbinti A, Myint KT. Work-related ocular injuries in Johor Bahru, Malaysia. International Eye Science. 2016;16(3):416-22.

- 79. Woh PY, Thong KL, Behnke JM, Lewis JW, Zain SN. Characterization of nontyphoidal Salmonella isolates from asymptomatic migrant food handlers in Peninsular Malaysia. Journal of Food Protection. 2017 Aug;80(8):1378-83.
- 80. Tanabe M, Myers A, Bhandari P, Cornier N, Doraiswamy S, Krause S. Family planning in refugee settings: Findings and actions from a multi-country study. Conflict and Health. 2017 Dec;11(1):9.
- 81. Ratnalingam V, Umapathy T, Sumugam K, Hanafi H, Retnasabapathy S. Microbial keratitis in West and East Malaysia. Internal Eye Science. 2017;17(11):1989-92.
- 82. Woh PY, Thong KL, Lim YA, Behnke JM, Lewis JW, Mohd Zain SN. Microorganisms as an indicator of hygiene status among migrant food handlers in Peninsular Malaysia. Asia Pacific Journal of Public Health. 2017 Oct;29(7):599-607.
- 83. Noor NM, Shaker MN. Perceived workplace discrimination, coping and psychological distress among unskilled Indonesian migrant workers in Malaysia. International Journal of Intercultural Relations. 2017 Mar 1;57:19-29.
- 84. Noordin R, Mohd Zain SN, Yunus MH, Sahimin N. Seroprevalence of lymphatic filariasis among migrant workers in Peninsular Malaysia. Transactions of The Royal Society of Tropical Medicine and Hygiene. 2017 Aug 1;111(8):370-2.
- 85. Sahimin N, Lim YA, Ariffin F, Behnke JM, Basáñez MG, Walker M, Lewis JW, Noordin R, Abdullah KA, Zain SN. Socio-demographic determinants of Toxoplasma gondii seroprevalence in migrant workers of Peninsular Malaysia. Parasites & Vectors. 2017 Dec 1;10(1):238.
- 86. Labao HC, Faller EM, Bacayo MF. 'Aches and pains' of Filipino migrant workers in Malaysia: A profile of work-related musculoskeletal disorders. Annals of Global Health. 2018;84(3):474.
- 87. Shaw SA, Ward KP, Pillai V, Hinton DE. A group mental health randomized controlled trial for female refugees in Malaysia. American Journal of Orthopsychiatry. 2018 Jul 23.
- 88. Rahman SA, Jeffree MS, Kamaludin F, Din IA, Yusof M, Ahmed K. Beriberi outbreak among unauthorised immigrants in a detention camp in Malaysia. Journal of Immigrant and Minority Health. 2018 Oct 1;20(5):1294-7.
- 89. Sahimin N, Douadi B, Lim AL, Behnke JM, Zain SN. Distribution of Giardia duodenalis (Assemblages A and B) and Cryptosporidium parvum amongst migrant workers in Peninsular Malaysia. Acta Tropica. 2018 Jun 1;182:178-84.
- 90. Nwabichie CC, Manaf RA, Ismail SB. Factors affecting uptake of cervical cancer screening among African women in Klang Valley, Malaysia. Asian Pacific Journal of Cancer Prevention. 2018;19(3):825.
- 91. Jeffree SM, Ahmed K, Safian N, Hassan R, Mihat O, Lukman KA, Shamsudin SB, Kamaludin F. Falciparum malaria outbreak in Sabah linked to an immigrant rubber tapper. The American Journal of Tropical Medicine and Hygiene. 2018 Jan 10;98(1):45-50.
- 92. Zerguine H, Tamrin SB, Jalaludin J. Prevalence, source and severity of work-related injuries among "foreign" construction workers in a large Malaysian organisation: A cross-sectional study. Industrial Health. 2018:2017-0205.
- 93. Ya'acob NA, Abidin EZ, Rasdi I, Rahman AA, Ismail S. Reducing work-related musculoskeletal symptoms through implementation of Kiken Yochi training intervention approach. Work. 2018 Jan 1;60(1):143-52.

- 94. Loganathan T, Rui D, Ng CW, Pocock NS. Breaking down the barriers: Understanding migrant workers' access to healthcare in Malaysia. PloS one. 2019;14(7).
- 95. Rahman MM, Arif MT, Safii R, Tambi Z, Akoi C, Jantan Z, Halim SA. Care seeking behaviour of Bangladeshi migrant workers in Sarawak, Malaysia. Bangladesh Medical Research Council Bulletin. 2019 Jun 15;45(1):47-53.
- 96. Siah PC, Low SK, Lee WY, Lim YY, Tan JT. Discrimination perceived by child refugees in Malaysia: from the views of representatives from refugees' community service centres and non-government service. Vulnerable Children and Youth Studies. 2019 Sep 1:1-4.
- 97. Sahimin N, Lim YA, Noordin R, Yunus MH, Arifin N, Behnke JM, Zain SN. Epidemiology and immunodiagnostics of Strongyloides stercoralis infections among migrant workers in Malaysia. Asian Pacific Journal of Tropical Medicine. 2019 Jun 1;12(6):250.
- 98. Chuah FL, Tan ST, Yeo J, Legido-Quigley H. Health system responses to the health needs of refugees and asylum-seekers in Malaysia: A qualitative study. International Journal of Environmental Research and Public Health. 2019 Jan;16(9):1584.
- 99. Vearey J, Orcutt M, Gostin L, Braham CA, Duigan P. Building alliances for the global governance of migration and health. BMJ. 2019 Sep 16;366:I4143.
- 100. World Health Organization Regional Office for Europe. Strategy and action plan for refugee and migrant health in the WHO European Region. Copenhagen: World Health Organization Regional Office for Europe 2016.
- 101. Reidpath DD, Allotey P. Infant mortality rate as an indicator of population health. Journal of Epidemiology & Community Health. 2003 May 1;57(5):344-6.
- 102. Vijayasingham L, Rhule E, Asgari-Jirhandeh N, Allotey P. Restrictive migration policies in low-income and middle-income countries. The Lancet Global Health. 2019 Jul 1;7(7):e843-4.
- 103. Juárez SP, Honkaniemi H, Dunlavy AC, Aldridge RW, Barreto ML, Katikireddi SV, Rostila M. Effects of non-health-targeted policies on migrant health: A systematic review and meta-analysis. The Lancet Global Health. 2019 Apr 1;7(4):e420-35.
- 104. Shao C, Meng X, Cui S, Wang J, Li C. Income-related health inequality of migrant workers in China and its decomposition: An analysis based on the 2012 China labor-force dynamics survey data. Journal of the Chinese Medical Association. 2016 Oct 1;79(10):531-7.
- 105. He X, Wong DF. A comparison of female migrant workers' mental health in four cities in China. International Journal of Social Psychiatry. 2013 Mar;59(2):114-22.
- 106. Al-Khatib IA, Ju'ba A, Kamal N, Hamed N, Hmeidan N, Massad S. Impact of housing conditions on the health of the people at al-Ama'ri refugee camp in the West Bank of Palestine. International Journal of Environmental Health Research. 2003 Dec 1;13(4):315-26.
- 107. Mutalip MH, Mishkin K, Paiwai F, Sulaiman J, Yoep N. Factors associated with sexual intercourse, condom-use, and perceived peer behaviors among adolescents in Malaysia: A school-based cross-sectional study. Malaysian Journal of Social Sciences & Humanities. 2019 May 1;4(2):8-18.
- 108. Khan MA, Rosly NA, Majeed AB, Ismail NE. Knowledge of modes of HIV transmission among public secondary school students in urban and rural areas in Malaysia. Pakistan Journal of Pharmaceutical Sciences. 2018 Sep 1;31(5).

- 109. Shakir M, Mahavera S, Wong LP, Lim Abdullah K, Adam P. Online STI information seeking behaviour and condom use intentions among young Facebook users in Malaysia. Health Promotion International. 2019.
- 110. Renzaho AM, Burns C. Post-migration food habits of sub-Saharan African migrants in Victoria: A cross-sectional study. Nutrition & Dietetics. 2006 Jun;63(2):91-102.
- 111. Ganle JK, Amoako D, Baatiema L, Ibrahim M. Risky sexual behaviour and contraceptive use in contexts of displacement: Insights from a cross-sectional survey of female adolescent refugees in Ghana. International Journal for Equity in Health. 2019 Dec 1;18(1):127.
- 112. Bosdriesz JR, Lichthart N, Witvliet MI, Busschers WB, Stronks K, Kunst AE. Smoking prevalence among migrants in the US compared to the US-born and the population in countries of origin. PloS One. 2013;8(3).
- 113. Ritchie H, Roser M. Smoking. Oxford: Our World Data; 2019 [updated 2019 Nov; cited 2019 Nov 20]. Available from: https://ourworldindata.org/smoking
- 114. Sweileh WM. Global output of research on the health of international migrant workers from 2000 to 2017. Globalization and Health. 2018 Dec 1;14(1):105.
- 115. World Health Organisation. Noncommunicable diseases (NCD) country profiles Malaysia. (n.p.): World Health Organisation 2018.
- 116. Zimmerman C, Hossain M, Watts C. Human trafficking and health: A conceptual model to inform policy, intervention and research. Social Science & Medicine. 2011 Jul 1;73(2):327-35.
- 117. Zimmerman C, Kiss L, Hossain M. Migration and health: A framework for 21st century policymaking. PloS Medicine. 2011 May;8(5).
- 118. Par M, Hassan SA, Uba I, Baba M. Perceived stress among International postgraduate students in Malaysia. International Journal of Psychological Studies. 2015 Dec 1;7(4):1.
- 119. Kohno A, Farid ND, Musa G, Aziz NA, Nakayama T, Dahlui M. Factors affecting Japanese retirees' healthcare service utilisation in Malaysia: A qualitative study. BMJ Open. 2016 Mar 1;6(3).
- 120. Asits S, Noorhidayah I, Osman A. Migrasi bermusim: Pengaruhnya terhadap kawalan malaria di Sabah. Medical Journal of Malaysia. 1999;54(2):200-209.
- 121. World Health Organization. A conceptual framework for action on the social determinants of health. Geneva: World Health Organization 2010.